



Comprehensive Review Based on the Impact of Integrating Electric Vehicle and Renewable Energy Sources to the Grid

Pampa Sinha¹, Kaushik Paul^{2,*}, Sanchari Deb^{3,*} and Sulabh Sachan⁴

- ¹ School of Electrical Engineering, KIIT University, Bhubaneswar 751024, India; pampa.sinhafel@kiit.ac.in
- ² Department of Electrical Engineering, BIT Sindri, Dhanbad 828123, India
- ³ School of Engineering, University of Warwick, Coventry CV4 7AL, UK
- ⁴ Electrical Engineering Department, GLA University, Mathura 281406, India
- * Correspondence: kaushik.ee@bitsindri.ac.in (K.P.); sanchari.deb@warwick.ac.uk or sancharideb@yahoo.co.in (S.D.)

Abstract: Global warming, pollution, and the depletion of fossil fuels have compelled human beings to explore alternate sources of energy and cleaner modes of transport. In recent years, renewable energy sources (RES) have been massively introduced to the grid. Furthermore, Electric Vehicles (EVs) are becoming popular as a cleaner mode of transport. However, the introduction of RESs and EVs to the grid has imposed additional challenges on the grid operators because of their random nature. This review aims to focus on the integration of RES and EVs to the grid, thereby presenting the global status of RESs and EVs, the impact of integrating RESs and EVs to the grid, the challenges of integrating RES and EV to the grid, optimization techniques for EV and RES integration to the grid, and mitigation techniques. A total of 153 research papers are meticulously reviewed, and the findings are put forward in this review. Thus, this review will put forward the latest developments in the area of EV and RES integration into the grid and will enlighten the researchers with the unsolved questions in the area that need investigation.

Keywords: electric vehicle; optimization; renewable energy source; power grid; soft computing

1. Introduction

Globally, pollution, fossil fuel depletion, and greenhouse gas emissions are the most alarming concerns. A paradigm shift from conventional energy sources to renewable energy sources (RESs) is observed. RESs are clean energy sources that emit no greenhouse gases and do not cause pollution. These RESs diversify the energy supply and decrease the dependence on imported fuels. Further, RESs provide avenues for jobs in manufacturing and installation, thereby enhancing economic development. Cheap electricity from RES could provide 65 percent of the world's total electricity supply by 2030. It could decarbonize 90 percent of the power sector by 2050, massively cutting carbon emissions and helping to mitigate climate change [1]. Similarly, EVs are becoming popular as a cleaner mode of transport. EVs are typically powered by a lithium-ion battery that is charged by electricity [2–5]. It is found that, with electricity generation, the carbon emissions of an electric car are around 17–30% lower than driving a petrol or diesel car [6]. However, integrating RESs and EVs into the network results in additional challenges for the grid operators. Some of the challenges associated with RESs are frequency and voltage anomalies, demand and supply mismatches, overloading of existing transmission lines [7–9]. This also includes some of the challenges associated with EV integration are peak load increases, voltage instability, degradation of reliability indices, and harmonic distortions [10–12]. This review aims to focus on the integration of RES and EVs to the grid, thereby presenting the global status of RESs and EVs, impact of integrating RESs and EVs to the grid, challenges of integrating RES and EV to the grid and the mitigation techniques, soft computing applications for EV and RES. Thus, this review will put forward the latest developments in the area of EV and



Citation: Sinha, P.; Paul, K.; Deb, S.; Sachan, S. Comprehensive Review Based on the Impact of Integrating Electric Vehicle and Renewable Energy Sources to the Grid. *Energies* 2023, *16*, 2924. https://doi.org/ 10.3390/en16062924

Academic Editors: Ahmad Hably, Reza Razi and Mehrdad Gholami

Received: 30 January 2023 Revised: 9 March 2023 Accepted: 14 March 2023 Published: 22 March 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). RES integration to the grid and will enlighten the researchers with the unsolved questions in the area that are worth investigating.

The remainder of the paper is organized as follows. Section 2 presents the existing reviews, highlights the contributions of the work, and elaborates on the search criteria. Section 3 presents the global statistics and research on RES and EV. Sections 4 and 5 present an overview of EV and RES integration into the grid, respectively. Section 6 presents the modeling and decision-making approaches in the case of EV and RES. Sections 7 and 8 present the research on the impact of integrating EV and RES to the grid, respectively. Sections 9 and 10 present the challenges of integrating EV and RES into the grid, respectively. Section 11 presents the optimization techniques applied for EV and RES integration into the grid. Section 12 discusses the key insights of the review. Finally, Section 13 concludes the work.

2. Existing Reviews, Contribution, and Search Criteria

2.1. Existing Reviews

The existing reviews on different aspects of EV and RES are listed in this section, as shown in Table 1. For example, in ref. [13], authors have meticulously reviewed 239 Q1 journals on the topic of EV adoption and put forward the factors affecting the global EV adoption rate. In ref. [14], the status of charging infrastructure development in the UK is reported. In ref. [15], the modeling of solar-based EV chargers is discussed. In ref. [16], data sources for EV-related research are presented. In ref. [17], the importance of smart charging strategies to tackle the increased load demand due to charging is reported. In ref. [18], the research works dealing with the interaction of EVs in a smart grid environment is reported. In ref. [19], the globally prevalent standards for EV charging are reported. In [20], EV-integrated Virtual Power Plants (VPPs) are reviewed in the context of benefits and applicability. In ref. [21], the applications of machine learning for solving the charging infrastructure planning problem are reviewed. In ref. [22], charger placement models are reviewed and compared. In ref. [23], charger placement in the context of smart cities is discussed. In ref. [24], technical and architectural requirements for RES integration to the grid is reviewed. In ref. [25], how RES can enhance the flexibility of power systems is discussed. In ref. [26], the concept and challenges of solar power integration into the grid are critically reviewed. In ref. [27], policies on RES integration into the grid are reviewed. In ref. [28], the effects and associated costs of integrating variable renewables into power grids are comprehensively reviewed. In ref. [29], the power quality challenges imposed by grid-integrated RES are discussed. In ref. [30], authors have reviewed RES integration impacts within the context of generator type, penetration level, and grid characteristics. Furthermore, in ref. [31], optimization techniques for integrating RES with the grid are reported. In ref. [32], the challenges of integrating wind energy into the grid are presented. In ref. [33], the necessity of stationary energy storage for RES integration into the grid is discussed. In ref. [34], applications of blockchain technologies for RES integration into the grid are discussed. In [35], authors surveyed load frequency control considering RES integration with the grid. In ref. [36], the pathways for RES-based electricity generation and supply are discussed.

Table 1.	Existing	Reviews	on EV	and RES	
----------	----------	---------	-------	---------	--

Reference	Year	Торіс	Diligence
[13]	2020	EV	Systematic presentation of 239 articles published in various Scopus Q1 journals related to EV adoption
[14]	2020	EV	Comprehensive review of charging infrastructure development in the UK
[15]	2020	EV	Review of PV based EV modelling approaches
[16]	2021	EV	Review of existing data sources for EV research
[17]	2022	EV	Review of smart charging strategies for EVs
[18]	2022	EV	Review of impact of EV charging on smart grid
[19]	2022	EV	Review of the standards and best practices for EV charging
[20]	2022	EV	Systematic review of EV integrated Virtual Power Plants

Reference	Year	Торіс	Diligence
[21]	2021	EV	Comprehensive review of machine learning applications for EV charging infrastructure planning
[22]	2021	EV	Comprehensive review of EV charging station planning models
[23]	2021	EV	Overview of charger planning models in the context of smart city
[24]	2019	RES	Review of architecture requirements for integrating RES to the grid
[25]	2020	RES	Survey of challenges of RES integration to the power system flexibility
[26]	2019	RES	Overview of solar power integration to the grid
[27]	2019	RES	Survey of research trends and policy implications of RES integration to grid
[28]	2021	RES	A systematic review of the VRE addition to power grids
[29]	2020	RES	Review of power quality challenges and state-of-the-art mitigation techniques of grid integrated RES
[30]	2019	RES	Review of RES integration impacts within the context of generator type, penetration level and grid characteristics
[31]	2020	RES	Review of optimization models for RES integration to the grid
[32]	2020	RES	Comprehensive review of grid integration challenges of wind energy
[33]	2022	RES	Review of stationary energy storage for RES integration to the grid
[34]	2022	RES	Review of applications of blockchain technologies for RES integration to grid
[35]	2022	RES	Survey on load frequency control considering RES integration with the grid
[36]	2022	RES	Review of sustainable RES electricity generation and supply

Table 1. Cont.

2.2. Key Contributions

The key contributions of this review as compared with the reviews listed in Table 1 are as follows:

- Review of the impact of EV and RES integration on power grid presented together
- Overview of global status of RES and EV integration
- Overview of optimization techniques for EV and RES integration to grid
- Review of challenges and mitigation strategies of EV and RES integration

2.3. Search Criteria

The search criteria followed for this review are as shown in Figure 1, thereby explaining the databases covered, the keywords used, and the exclusion criteria.



Figure 1. Search Criteria followed in the Review.

The statistics related to different aspects of EV and RES, as well as the global research on EV and RES, are presented systematically in this section.

3.1. EV Statistics

The global statistics related to EV sales, shares, and chargers are presented in this section. It must be noted that these statistics are collected from global database, such as refs. [37–40].

Figure 2 presents the EV registration and share for the year 2021 for different countries such as Japan, Korea, Canada, United Kingdom, France, and Germany. It is observed that Germany has the highest EV share at 26%. Figure 3 shows the sales of different models of EV for the year 2022. It is observed that the Tesla Model Y was the highest-selling EV for 2022. Figures 4 and 5 show the year wise availability of fast and slow public chargers across different countries, respectively. Figure 6 shows the global distribution of EVs per charging point for the year 2021. It is observed that Norway has the most EVs per charging point.



Figure 2. EV registrations and share statistics for 2021 [37].



Figure 3. Sale of different EV models in 2022 [38].



Figure 4. Global availability of fast public chargers [39].



Figure 5. Global availability of slow public chargers [39].



Figure 6. Global distribution of EV per charging point for 2021 [40].

3.2. RES Statistics

The global statistics related to RES generation are reported in this section. It must be noted that these statistics are collected from global databases, such as ref. [41].

Figure 7 shows the variation in renewable energy generation across different countries from 2000 to 2021. It is observed that China has the highest annual change in renewable energy generation at more than 600 TWh. Figure 8 shows the annual change in solar energy generation across different countries. It is observed that China has the highest annual change in solar energy generation at more than 160 TWh. Furthermore, Figure 9 shows the annual change in wind energy generation across different countries. It is observed that China has the highest annual change in wind energy generation across different countries. It is observed that China has the highest annual change in wind energy generation across different countries. It is observed that China has the highest annual change in wind energy generation at more than 400 TWh.



Figure 7. Annual change in renewable energy generation across different country [41].



Figure 8. Annual change in solar energy generation across different country [41].



Figure 9. Annual change in wind energy generation across different country [41].

3.3. Global EV Research

The global trends in research on EV and its related aspects are comprehensively reported in this section.

In ref. [42], Kazemzadeh et al. examined the influence of BEVs and PHEVs that can reduce the release of fine particulate matter (PM2.5) with the application of the moment's quantile regression (MM-QR) which has been investigated considering a panel of 29 European nations from 2010 to 2019. The research work examined the association between the rising usage of electric cars and PM2.5 emissions. Their study introduces a new approach, the MM-QR, to examine the correlation between the increasing adoption of electric vehicles and the release of PM2.5. Two models were developed in order to assess their impact on PM2.5 reduction in European nations. The models' nonlinearity was established. The statistical evidence of the parameters was confined to the higher quantiles (75th and 90th), owing to the efficacy of European environmental legislation. Electric cars (BEVs and PHEVs), economic development, and urbanization all help to lower PM2.5 emissions, but fossil fuel usage exacerbates the issue. This study gives insight on how governments and policymakers in Europe may devise plans to increase the usage of electric vehicles.

In [43], Suttakul et al. compared the TCO of EVs to that of traditional ICE vehicles. The authors considered HEV, PHEV, and BEV in the analysis. Cost models were developed considering both capital and operational expenses, including energy consumption, loan interest, the cost of depreciation, upkeep, and taxes. The complete data analysis has been derived from field tests conducted in Thailand. The proposed TCO models were examined utilizing an average urban driving distance of 20,000 km per year for a period of 15 years. ICE, HEV, PHEV, and BEV have a total cost of ownership (TCO) of 61.19, 54.94, 55.55, and 60.89 per \$1000 USD, respectively. The research also analyzed the lifetime TCO ratio for each ownership year. It was found that in Thailand, the HEV and PHEV are viable options without BEV support regulations. In addition, the study applies the TCO framework and presents the results as a TCO ratio over a 15-year period, evaluating two hypothetical scenarios where different EV support measures are implemented: either a government subsidy or a discount at the point of purchase, and a reduction in battery prices. The total

cost of ownership (TCO) is greatly reduced through direct assistance for the purchase of cars. Government authorities, manufacturer product planners, and consumers might benefit from the information gleaned from this research.

In ref. [44], Nadeem et al., analyzed the CO_2 emission reductions, fuel consumption, and future power demand in Pakistan as a result of the anticipated increase in the number of EVs. The ePop model, which forecasts the future energy consumption of EVs, is used to plan for EVs until 2040. The study evaluates two scenarios: one where 30% of all electric cars and vehicles are in use by 2030, and another where 90% are in use by 2040, both compared with the number of electric vehicles in use in 2020. Consideration is given to PV during the day and the national electric grid at night to meet the anticipated energy consumption. To supply the energy needed for the EVs' consumption of 14.7 TWh/year by 2040, 9 GW of PV capacity will be necessary. On the other hand, 0.7 GW of power plant capacity will be required to fulfill the demand of 4.7 TWh/year. Furthermore, daytime and overnight charging situations for electric vehicles are evaluated. The results predicted a decrease of 10.4 MtCO₂ emissions and 9.1 Mtoe of fuel usage in the transportation industry by 2040.

In [45], Wesseh et al. used a game-theoretic method to analyze the interaction between pumped hydro storage, electric vehicles, and climate policy within the electricity market framework. The study focused on the collaboration between the ownership of storage assets and power generation, taking into account various market power dynamics. Chinese electricity market data has been considered to analyze this framework, and it is shown that climate policy may not encourage storage use, especially when companies that hold a dominant position in the storage industry have a limited portfolio of power generation assets. Moreover, when a perfect competitive market is considered, EVs may not increase the welfare of customers in the presence of pumped hydro storage. These findings have a substantial impact on the policy ramifications.

In ref. [46], Li et al. proposed mathematical models for conducting economic analyses of the viability of hydrogen energy generated from RES and then implementing them in the Chinese road transportation industry. The development of a well-to-wheel model to assess the carbon emissions of the hydrogen supply chain and fuel cell EVs. In the interim, a levelized cost of hydrogen model has been formulated to examine the cost of hydrogen as a sustainable energy storage medium. Furthermore, an aggregated cost of ownership model is implemented to estimate the owing cost of and operating a fuel cell EV, powered by hydrogen generated from RES, in comparison to other vehicle powertrains, particularly those powered by fossil fuels. On this basis, the relationship between energy policy and the competitiveness of hydrogen derived from RES and fuel cell EVs is examined.

In ref. [47], Asadi et al. aimed their study at analyzing the elements that influence customers' propensity to utilize EVs. The researchers developed a model based on the Norm Activation Model and the Theory of Planned Behavior to understand the factors that influence people's desire to purchase electric cars. They conducted a survey of potential customers in Malaysia and collected 177 valid responses, using a structural equation model to analyze the data. The results showed that perceived value, attitude, responsibility attribution, subjective norms, personal norms, perceived consumer efficacy, and knowledge of consequences all had a significant and positive impact on customers' willingness to buy EVs.

In ref. [48], Schulz et al. investigated how public charging infrastructure influences the uptake of BEV. Our research has been based on yearly, detailed data on the placement of public charging infrastructure and the ownership rate of battery electric vehicles in 356 Norwegian LAU-2 towns from 2009 to 2019. They concentrated on regions where the initial infrastructure for public charging has been deployed within this time frame. In these mostly rural locations, proliferation began with the installation of the first public charging station. They observed that there has been a rise of 1.5 percentage points, or 200 percent in the local EV ownership rate over five years. In addition, the research outcomes are unaffected by other treatment criteria, such as the median number of public chargers in a region from 2009 to 2019 or the median density of public charging outlets

per 1000 residents over the same time period. Although their research cannot rule out the possibility of reverse effects, they identified public charging stations as a stimulant for the spread of BEV.

3.4. Global RES Research

The global trends in research on RES and its allied aspects are comprehensively reported in this section.

In ref. [49], Toklu et al. demonstrated in their research that Turkey has a tremendous potential for renewable energy, particularly hydropower, biomass, geothermal, solar, and wind. The renewable energy technologies of wind, biofuels, solar thermal, and photovoltaics are now maturing and offering the ultimate promise of cost-competitiveness. Environmental pollution is turning into a major worry for the nation as the consumption of energy and power in Turkey increases fast and becomes more reliant on imports of costly fossil fuels that throw a heavy weight on the economy. Concerning worldwide environmental challenges, Turkey's carbon dioxide emissions and energy usage have both increased. States have taken the initiative in safeguarding the environment by lowering the emissions of greenhouse gases. In this sense, RES seems to be one of the most effective and efficient alternatives for Turkey's growth in clean and sustainable energy.

In ref. [50], Poudyal et al. indicate that RES are vital not only for reducing the current energy shortage, but also for achieving energy sovereignty for Nepal through the establishment of dependable and sustainable energy sources. Their research attempted to present an up-to-date view on Nepal's present energy problem, taking into account the dynamic nature of Nepal's energy status and the latest advancements in renewable energy technology. Specifically, the existing profiles of energy production and consumption are analyzed, and the primary reasons leading to the expanding imbalance between energy supply and demand are highlighted. These aspects include deferred and overly expensive hydropower projects, obsolete and inadequate energy infrastructure, transmission and distribution losses, energy theft, inadequate energy governance, a lack of energy sustainability, low equipment effectiveness, inefficient energy pricing models, and inadequate energy market regulations. Specific geographical and geopolitical issues, a large reliance on energy imports, and insufficient use of the huge quantities of renewable energy resources are also significant contributors to the escalation of the energy crisis. The current and future status of significant hydroelectric projects is summarized. Recent policies and investment efforts of the Nepalese government to promote green and sustainable energy are examined. In addition, a long-term prognosis on the energy situation in Nepal is provided using the energy modeling program LEAP to demonstrate how to use Nepal's vast renewable energy resources.

Reference [51], authored by Shakeel et al., proposed that decision-makers and policymakers in Pakistan should have knowledge of energy innovation and resource expansion. They suggested prioritizing certain energy sources and addressing short- and long-term concerns when making decisions. Their research presented a detailed description of the Pakistani power industry and the problems it faces. Moreover, an examination of the country's energy policies throughout the years and their influence on the power industry is provided. The study concluded that Pakistan's current energy profile is not sustainable due to the country's heavy reliance on imported fossil fuels, which leads to increasing electricity generation costs and rising emissions associated with power production. The study presented a clear roadmap and suggested energy sources that may both meet the nation's expanding energy demands and be environmentally responsible. The strategic planning outlined and emphasized the most important steps the nation must take to realize its aim of fulfilling its energy demands and incorporating RES into its power production.

In ref. [52], Ślusarczyk et al. examined the reciprocal relationships between RES and economic development for two European Union countries, as well as the impact on their variations (increase and decrease). A quantitative examination of outcomes was conducted for the economies of low-income Poland and high-income Sweden. This study employed a regression model to examine the existence of relationships between RES, gross incremental

energy usage, and economic development. Their research examines data from 1991 until 2022. The findings demonstrated a favorable connection (statistically significant) between the Gross Domestic Product and Gross National Income factors that impact the usage of RES in Sweden (84.6% and 83.7%, respectively) and Poland (79.9% and 79.9%, respectively). The data also indicated that the usage of RES has resulted in a better economic growth rate in the leading nations, but the danger of recession is significantly higher than in other nations. These results would aid state officials and policymakers in comprehending the importance of RES in the economic development of these nations.

In ref. [53], Amhed et al. found that in the future, renewable energy solutions with the lowest operating and externality costs will be the best option. From a policy standpoint, the Pakistani government should support RES and technical development, which requires biomass resources to be connected to non-renewable, protracted expenditures. However, officials have defined several programs to meet energy demand, but they are still unable to close the supply-demand imbalance. 11% of the global population has no access to alternative energy supply and access methods. Moreover, unique strategies for the development of renewable energy have emerged at various times. It covers homes in particular in remote locations without access to gas or electricity. Nonetheless, the aim of this study is to determine the most significant renewable energy source for Pakistan's economy, considering economic benefits such as job creation in the energy sector. This research tries to discover methods for securing energy supplies and generating economic advantages.

In ref. [54], Pereira et al. demonstrated the prospects for the incorporation of new RES into Brazil's energy mix, therefore allowing the country to continue producing a greater proportion of clean energy than the global average. Within the 2010–2030 timeframe, several evolution models for such sources in Brazil and the rest of the globe were evaluated. The analysis revealed not only the advantages offered by any of these sources in the form of GHG emission minimization but also the effect in terms of job creation and the public expenditure required to get these benefits.

4. Overview of EV Integration to Grid

Electrification of transport is one of the key initiatives towards achieving the decarbonization goals. EVs need to be integrated into the grid during charging, and further, EVs can serve as a source by discharging or giving power back to the grid during peak load periods. Different aspects of integrating EVs into the grid, such as EV behavior modeling and the siting as well as sizing of BSS, are discussed in this section.

4.1. EV Behavior Modelling and Applications

Modeling of EV usage behaviors primarily concerns traveling, parking, and charging activities. Modeling of EV charging behavior is essential for predicting the charging demand and scheduling the charging activities without compromising with the secure operation of power grid [55–60]. The demand for charging an EV depends on trip number, trip distance, energy consumption, and the availability of charging infrastructure. The availability of charging infrastructure is further characterized by parking location, permitted charging power, and intended parking time. EV charging behaviors can be identified from charging profiles or travel and parking activities. The different types of EV usage models are shown in Figure 10.

Temporal usage of EVs indicates the start and end times of different EV usage events, including traveling, charging, and parking, based on which the duration and frequency of each event can be obtained. The temporal usage can be further sub divided into a statistical model, a queuing model, and a time Markov chain model.

EV spatial usage refers to the spatial information of EV activities, including trip destinations, charging locations, and other socio-demographic point of interests (POIs). The spatial usage can be determined by origin destination analysis or trip chain analysis.



Figure 10. EV Usage models.

Energy usage modeling is also important for EV charging demand prediction, as temporal and spatial usage models are unable to obtain charging demand profiles without knowing the energy consumption of EVs. This can be again travel energy consumption or charging energy consumption.

A summary of the EV usage pattern models is presented in Table 2.

Table 2. Summary of EV usage pattern models [61].

Model	Sub Model	Description	Application
	Statistical Model	Statistics/distributions of start/end time/duration	Inputs for charging demand simulation and other temporal usage models
Temporal Usage	Queuing model	EV arrival rate and the numbers of charging EVs at assigned charging locations	Charging demand analysis of preassigned charging locations
	Time Markov Chain model	Transition probability matrices of vehicle state	Generate sequences of EV state over a period
Spatial Usage	Origin Destination analysis	Origin Destination matrices	Simulate spatial movements and trip destinations
	Trip chain model	Series of trip destinations and PDFs of departure/arrival time	Generate commuting patterns for multi-location demand estimation and charging scheduling
Energy Usage	Travel energy consumption	Travel distance, energy consumption rate	Calculate electricity demands of EVs
	Charging energy consumption	Charging power, charging time, charging efficiency	Charging power, charging time, charging efficiency

4.2. Siting and Sizing of EV Battery Swapping Station

Building BSS is very expensive. Therefore, a reasonable siting and sizing can not only provide excellent accessibility to battery swapping service for consumers, but also save a large amount of construction and configuration costs for BSS operators. The main factors affecting BSS siting and sizing are the technical ability of BSSs and the EV usage distribution and battery swapping habits of EV users. Service radius, traffic congestion and power capacity of local distribution network should be considered when determining the location and capacity of a BSS. An overview of the siting and sizing framework of BSS is shown in Figure 11.



Figure 11. Siting of BSS [62].

5. Overview of RES Integration to Grid

Numerous environmental concerns have led to an uptick in the interest in RES such as solar and wind energy. However, when it comes to recharging zero-emission automobiles, wind power is by far the most popular option. This is an essential application of wind power.

According to the global wind report published by the GWEC in 2021, the objective for power system engineers in the future decade will be to create wind power plants with a greater capacity for production in order to mitigate the worst effects of climate change while simultaneously meeting demand. In the next decade, the entire global electricity production will be replaced by renewable energy, primarily wind power generation. Wind energy will provide a quarter of the non-conventional energy sources' overall output. As the utilization of wind power increases, conventional energy consumption will decrease, resulting in a greener environment. Consequently, greenhouse gas emissions can be decreased. Now is the moment for all of us to prioritize the utilization of renewable energy. Among all types of renewable energy, also known as clean energy, wind energy is currently gaining popularity due to its eco-friendly characteristics. However, the penetration level of wind power into a power system is difficult due to its randomness and uncertainty. The random character of the variance in wind power is a challenge for frequency management [63] and voltage quality [64].

6. Modelling and Decision Making

Compared with other forms of energy generation, wind energy generation is unpredictable, highly uncertain, and intermittent. Therefore, an efficient model is necessary for forecasting wind power. It must be as precise as feasible in order to reduce the consequences or repercussions of wind input on the grid. Only when EVs are parked can they be regulated; consequently, for load scheduling purposes. Awareness of their driving intentions is required. Human society is dynamic; hence, it is difficult to foresee its future. The moving information of an EV comprises the arrival time, departure time, and driving distance. As there is a deadline to complete charging for every electric vehicle parking action, this procedure of charging the EV is time-sensitive. The future cost will depend entirely on the current pricing selection. In a multi-stage setup, we must therefore schedule the EV charging load. And, as the quantity of EVs increases, the curse of dimensionality issue manifests itself.

As detailed in the comprehensive identification of oysters with superior qualities that include unpredictability [65], the rigorous scheduling strategy is crucial for the current circumstance [66]. Wind energy's penetration into the electrical system [67] and the integration of conventional power plants [68]. In their respective literatures, the rigorous scheduling between charging electric vehicles and wind power is rarely described for clarity. Then, several authors formulated this topic as a stochastic problem that considers the performance strategy in a wind energy crisis [69]. The RSP identified a robust strategy to reduce the total cost in the most uncertain future scenarios. In their development, a probabilistic and parametric density function is used to characterize the wind power. In the preceding description, it was assumed that the parameter was perfectly familiar [70]. Currently, some have estimated the required parameter range, which is more pragmatic. A full description is provided. RSSP is an exceptional approach to the Markov decision process (MDP), in which the achievement of a process is the standard price over the unconditional development contingency [71]. When different levels of the unconditional development condition are autonomous [72]. When the uncertainties of the transition probabilities at distinct stages are independent, also known as the rectangularity property, the robust MDP can, in principle, be solved by robust value iteration and robust policy iteration [73]. However, the computational complexity of these approaches typically renders their employment in large-scale situations fruitless. Thus, proximal problem-solving techniques are in significant demand [74]. There are further methods [75], such as a rigorous scheduling mechanism that enhances the unconditional features of a chosen standard strategy.

In 2018, robust scheduling is considered in terms of the following results: (i) formulation of scheduling as a robust stochastic shortest path, where the goal aim is a sum of wind power utilization and cost of charging. (ii) enhancing the technique based on simulation technique for inclement weather (iii) the performance is quantitatively presented using actual wind and electric car data as a foundation and presenting an RSSP model for load supply matching to reduce charging costs and improve demand and supply balance. For various conditions, simulations and mathematical calculations are performed [76].

7. Impact of EV Integration to Grid

Table 3 summarizes the results of EV integration with the grid. The effect of EV charging on the ToU rates of the electricity distribution network was studied by the authors of [77]. The net load demand of the power grid increases due to recharging electric vehicles. ToU pricing is a practical method to move EV charging away from the highest demand time of day. Introducing ToU pricing has been shown to have a measurable effect on EV charging habits, and this effect has been quantified by the authors. The simulation results showed that the peak load may be lowered by 5% by using ToU rates. The current harmonic distortion caused by recharging electric vehicles is reduced from 4.88 percent to 4.03 percent thanks to a compensation-based harmonic reduction technique provided in [78].

Table 3. Research work on impact of EV on power grid.

Ref	Author	Year	Parameter	Diligence
[77]	Jones et al.	2022	Peak load	ToU rates reduced the peak load by 5%
[78]	Kazemtarghi et al.	2022	Harmonics	Active compensation based harmonic reduction technique that reduced the EV charging induced harmonic distortion in current from 4.88% to 4.03%

Ref	Author	Year	Parameter	Diligence
[79]	Singh et al.	2022	Voltage profile, Transformer loading	Impact of uncontrolled, smart, and V2G strategy on a residential power distribution network of Netherland
[80]	Qiu et al.	2022	Residential load demand	Empirical estimation of the grid impact due to in-home EV charging was different from that predicted by existing simulation models due to factors such as consumer behaviors
[81]	Pankaj et al.	2022	Harmonics, Voltage profile	With 1 EV, 20.30% of THD is produced, and with 3 EVs, 27.67% of THD is produced. Low voltage is also observed when more than one EV charger is connected at one phase
[82]	Alquthami et al.	2022	Load demand, power losses, voltage profile	Innovative model to generate the profiles of EV charging demand of Saudi Arabia and investigate the impact of charging on grid
[83]	Ding et al.	2022	Reward due to discharging	Bottleneck model considering trade off between waiting time cost and reward due to discharging
[84]	Rancilio et al.	2022	Tariff, Peak power, carbon footprint	By adopting smart charging at home the EV evening peak load can be reduced by 30% to 50%. Further, a 10% decrease in carbon footprint is achieved by valley-filling with work charging
[85]	Rahman et al.	2022	Voltage profile, harmonics, stability, power quality	EVs were modelled as mobile loads and was used as flexible resources
[86]	Mane et al.	2022	Transformer loading, Power factor, phase asymmetry, voltage profile	Modelled the impact of EV charger on LV residential network. Generally, nonlinear EV loads are connected to different phases, so there will be a phase asymmetry in the network. This asymmetry can be reduced by proper grid planning. Turning on and off of EV charger may impact the power factor that can be tackled by reactive power compensation
[87]	Deb et al.	2022	Voltage profile, power quality	Placing the EV load on weak points of the power distribution network and uncoordinated charging may be detrimental to smooth operation of the grid
[88]	Zaferanlouei et al.	2022	Voltage profile, transformer loading, congestion	Overloading of lines and transformers would take place when the share of EVs is above 20% for Norwegian grid
[89]	Sachan et al.	2021	Load demand, waiting time in the charging station	Modelled the charger placement problem by a game theoretic approach and analyzed the impact of charger placement on load demand and waiting time in the charging stations
[90]	Dechanupaprittha et al.	2021	Load demand, frequency deviation	Self-learning PSO based optimal EVs charging power control strategy considering the net charging power demand and frequency deviation to ensure smooth functioning of the grid
[91]	Ahmad et al.	2021	Harmonics, thermal limit, peak load	The voltage profile of the buses of the distribution network of Qatar was examined as the EV load increased to determine the capacity of the buses at certain THD levels
[92]	Khan et al.	2021	Voltage sag, load demand, power quality	Residential grid voltage sag increased by about 1.96% to 1.77%, 2.21%, 1.96 to 1.521% and 1.93% in four EV-charging profiles, respectively

Table 3. Cont.

Ref Author Year Parameter Diligence An open-access agent-based EV simulation model was proposed including a probabilistic [93] 2021 Peak load, charging frequency plug-in decision module developed and Venegas et al. calibrated to match the charging behavior observed in the Electric Nation project Simulated results showed that compared to the non-dynamic charging mode, the electric bus [94] Qiu et al. 2021 Reliability dynamic charging mode does not cause additional deterioration to the reliability indices such as SAIFI, SAIDI, and AENS Simulation results considering 3 million passenger EVs on the road showed that the [95] Mowry et al. 2021 Peak load, operational cost system operational cost would rise by 8%. The rise in operational costs would be mostly due to transmission congestion of the feeder lines Transformer loading, line Grid operating parameters are most sensitive to [96] 2021 Stiasny et al. loading, peak load, voltage number of vehicles, charger rating, and driver profile behavior modelling Modelled the charger placement problem by Chicken Swarm Optimization (CSO) based Voltage stability, reliability, [12] Sachan et al. 2021 heuristics and further analyzed the impact of power losses charger placement on voltage stability, reliability indices, and power losses Second Order Cone programming (SOCP) Voltage deviation, power formulation of the AC optimal power flow with [97] Soofi et al. 2021 quality the inclusion of EV and solar PV as distributed resources Analyzed the impact of EV charging on current harmonic emission, transformer loading, and Current harmonic emission, voltage distortions of a low voltage residential transformer loading, and [98] Iqbal et al. 2021 grid. Further, they put forwarded a Monte Carlo voltage distortions simulation-based approach to model the EV usage Harmonics, voltage variations and flicker were within the limits as given in EN50160. However, Slangen & Harmonics, Power quality, [99] 2021 the chargers were a source of superharmonic Bhattacharyya voltage profile currents at some specific frequencies. The impact on the grid voltage differed as per frequency Modelled the charger placement problem with and without V2G considering a novel index [100] Archana & Rajeev 2021 Reliability called EV placement index that takes into account the voltage stability, reliability, as well as cost associated with EV placement Coordinated charging strategy for the EVs [101]Akil et al. 2021 Peak load, line loading considering their dynamic behavior Predicted the future EV charging demand of an [102] Van den et al. 2021 Transformer loading office complex in order to estimate the flexibility potential of EVs Simulations showed that in urban networks EV Power quality, voltage profile, Haider & [103] 2021 integration would cause higher cable loading and cable loading Schegner in rural networks it would cause voltage drops

Table 3. Cont.

Ref	Author	Year	Parameter	Diligence
[104]	Sharma et al.	2020	Power quality	Comparative analysis of how common DC and AC bus architectures of fast chargers would impact the power quality of the grid
[105]	Sachan et al.	2020	Voltage profile, power quality	Compared different charging strategies and ranked smart charging highly to tackle the adverse impact of EV charging on grid
[106]	Deb et al.	2020	Voltage stability, reliability, power losses	Novel approach for charger placement considering cost, accessibility index, and VRP index
[107]	González et al.	2019	Peak load, voltage profile	Modelled the impact of EV charger on a Latin American grid and concluded that with 10% EV penetration the operational parameters of the grid would be least affected
[108]	Lillebo et al.	2019	Voltage profile, peak load	Simulations showed that the weakest power cable in the system would experience overload at a 20% EV penetration level. The network tolerated an EV penetration of 50% with regard to the voltage levels at all end-users
[109]	Deb et al.	2019	Voltage stability, reliability, power losses	Multi-objective formulation of the charger placement problem for Guwahati city considering cost, reliability, power
[110]	Fischer et al.	2019	Peak load	Stochastic bottom-up approach to analyze how EV charging would affect the load profiles
[111]	Deb et al.	2018	Voltage stability, reliability, power loss	System could sustain placement of fast charging stations at the strong buses up to a certain level, but the placement of fast charging stations at the weak buses of the system hampers the smooth operation of the power system
[112]	Kongjeen & Bhumkittipich	2018	Voltage profile	The lowest increased value of EV charging load had an effect on the load voltage deviation (0.062), the total active power loss (120 kW) and the total reactive power loss (80 kVar), respectively
[113]	Sachan	2018	Voltage profile, peak load	Smart charging strategies are beneficial and could contribute to less voltage violations and cost savings
[114]	Lin et al.	2018	Peak load	Proposed a multi-agent system to simulate the interaction of EVs in energy hub with different penetration rates and charging pattern
[115]	Deb et al.	2017	Reliability	After placement of fast charging station with 30 servers at bus 11 which was the strongest bus of the system the SAIFI increased to 0.1080 interruption/yr. SAIDI and CAIDI value also increased to 0.5374 h/yr and 5.2481 hr/interruption respectively

Table 3. Cont.

The effect of electric vehicle chargers on the voltage distribution network in a home was investigated in [79]. Evidence from public chargers in the Netherlands was used to model the charging process. The analysis of the consequences was done in three different cases. Uncontrolled charging was studied first, followed by smart charging, and finally bidirectional V2G. The simulations showed that the network performance metrics were suitable for a high density of EVs per charger. Nevertheless, an increase in the number of electric vehicles using a single charger will cause long wait times and irritate EV owners.

In addition, it was discovered that clever charging procedures will raise the transformers' maximum loading.

The authors of [80] looked at the environmental effects of home EV charging in Arizona, USA. In this study, we looked at information from the smart meters of around 1600 residences that have electric vehicles. The demand for residential load was shown to rise by 7–14% if EV charging occurred during summer peak hours (6 PM to 8 PM). It was also noted that most EV households responded to energy price signals by boosting their charging activity during off-peak hours when the ToU pricing was at its most affordable for EVs. Furthermore, consumer behavior and other factors accounted for discrepancies between existing simulation models and the actual estimation of the grid impact due to in-home EV charging.

To determine how EV charges affect harmonics, the writers analyzed the data presented in ref. [81]. According to the authors, harmonics rise as the number of EV charges grows. One EV results in 20.30% of THD, while three EVs result in 27.56% of THD. When more than one electric vehicle charger is plugged into the same phase, the voltage drops.

The authors of ref. [82] suggested a novel methodology to construct the profiles of EV charging demand that takes into account the driving patterns of EVs, energy consumption, and charging schedules. To test how charging electric vehicles would affect the electricity grid, a simulation was run using the suggested EV demand model. Saudi Arabia's distribution system was used to verify the model.

The authors of ref. [83] offer a bottleneck model to examine the effects of the bidirectional discharging by V2G method on waiting times and traffic congestion. The model also took into account the potential loss owing to waiting time versus the potential gain due to discharge.

The influence of smart charging on tariffs, peak load reduction, and carbon footprint reduction is investigated in ref. [84]. They concluded that 30% to 50% of the nighttime peak load for EVs might be avoided if more people used smart charging at home. Furthermore, through valley-filling with work charging, a 10% reduction in carbon footprint can be accomplished.

The effects of EV charging loads on power grid parameters such as voltage profile, stability, power quality, and harmonics were investigated by the authors in [85]. Opportunities for EVs to be used as a flexible resource were also explored, and EV loads were modeled as mobile loads.

The authors of ref. [86] looked into the viability of the current system for accommodating EVs. Grid factors such as transformer loading, power factor, voltage profile, and phase imbalance were examined to see how they might be impacted by the increased loading caused by EV charging. The authors have simulated how EV chargers will affect the LV home network. There will be phase asymmetry in the network because nonlinear EV loads are usually connected to separate phases. A strategic grid layout can mitigate this disparity. The power factor may be affected by the frequent on-and-off cycles of an electric vehicle charger; however, this is something that may be addressed through reactive power correction.

Using data from a sample of India's urban distribution networks, the authors of [87] studied how EV charging loads altered the voltage profile and power quality. They came to the conclusion that uncoordinated charging and concentrating the EV load on power distribution network weak spots could be harmful to grid stability.

Several charging scenarios, including "dumb" charging, "coordinated" charging without network con-straits, and "coordinated" charging with network limitations, were examined for their effects on the Norwegian grid by the authors of ref. [88]. It was determined that if the percentage of EVs on the road increased above 20%, there would be over-loading of wires and transformers.

The authors of ref. [89] modeled the charger placement problem using a game-theoretic method and investigated the effect of charger placement on load demand and waiting time at charging stations. Guwahati, a city in northeastern India, was used to verify the model's accuracy for its distribution network.

In order to keep the power grid running smoothly, the authors of ref. [90] developed an optimal charging power control method for electric vehicles (EVs) that uses a self-learning PSO to consider both the net charging power demand and the frequency deviation.

The effects of electric vehicle (EV) charging on Qatar's distribution network were studied in ref. [91]. Several scenarios of electric vehicle penetration into the distribution network were modeled in Simulink. With the goal of establishing the buses' capacity at a given THD level, the voltage profile was analyzed as the EV load rose.

By considering factors including charging duration, charging technique, and vehicle characteristics, the authors of [92] were able to assess the effects of widespread EV penetration on low voltage distribution. Observations made during the rapid charging and discharging of EVs showed that bus voltage and line current were impacted. In four EV-charging profiles, the residential grid voltage sag increased by 1.96%, 1.77%, 2.21%, and 1.96 to 1.521%, respectively.

In [93], the authors investigated the potential of EV fleets as a flexibility resource by analyzing the impact of both regular and irregular (not charging everyday) charging behavior on the grid. The Electric Nation project inspired the proposal of a new type of open-source agent-based EV simulation model, complete with a probabilistic plug-in decision module designed and calibrated to mimic real-world charging patterns. The average number of times an EV user plugs in their vehicle each week was calculated to be between two and three. The proposed agent-based model's results demonstrated that unlike when EVs are charged concurrently, when users aren't being systematic about plugging them in, the impact of EV charging is reduced, especially when charging in response to price. Non-systematic plug-in, however, can limit options, especially in light of recent trends toward ever-larger battery capacities.

The effects of dynamic wireless charging of electric buses on the power grid were studied by the authors of ref. [94]. In order to study how charging electric buses wirelessly will affect the power system, a dynamic model was developed. The dynamic charging mode of the electric bus was found to have no negative impact on dependability indices such as SAIFI, SAIDI, and AENS in simulations compared with the non-dynamic charging mode.

There was an examination of how highway fast chargers affected the electricity grid in ref. [95]. The operational cost of the system was predicted to increase by 8% in the simulation with 3 million passenger EVs on the road. The main cause of the increase in operating expenses would be transmission congestion on the feeder lines.

In ref. [96], authors modeled how recharging EVs might affect parameters of the low voltage distribution grid such as line loading, transformer loading, peak load, and voltage profile. The authors also conducted a sensitivity analysis and found that the number of vehicles, charger rating, and driver behavior model-ling have the greatest impact on grid operating parameters.

The authors of ref. [12] modeled the charger placement problem using heuristics based on Chicken Swarm Optimization (CSO), and then they examined the effect of charger placement on voltage stability, reliability indices, and power losses.

Rapid EV charging was studied for its effect on the IEEE 33 bus test network in ref. [97]. They reasoned that the voltage profiles of the buses would be affected by the charging levels. To further incorporate EVs and solar PV as distributed resources, they suggested a Second Order Cone Programming (SOCP) formulation of the AC optimum power flow.

In ref. [98], researchers examined how charging electric vehicles affected the current harmonic emission, trans-former loading, and voltage distortions of a low-voltage residential grid. In addition, they proposed a Monte Carlo simulation-based method for modeling EV utilization.

The authors of ref. [99] studied data from a field trial of electric bus charging that took place in real time. After careful analysis, they determined that the levels of harmonics, voltage fluctuations, and flickering were all well within the standards set out by EN50160. Yet, at some frequencies, the chargers produced currents that were themselves super harmonic. Grid voltage was affected differently depending on frequency. In ref. [100], authors modeled the charger placement problem with and without V2G considering a novel index called EV placement index that takes into account the voltage stability, reliability, as well as cost associated with EV placement. Further, the authors have simulated how the placement of chargers in the real time distribution network of Kerala, India, would impact the typical power system operating parameters such as voltage stability, harmonics, and reliability indices.

In ref. [101], authors proposed a coordinated charging strategy for the EVs considering their dynamic behavior. They also quantified the impact of the proposed strategy on power system parameters and concluded that a coordinated charging scheme has a lesser impact on the power network.

In ref. [102], authors focused on predicting the future EV charging demand of an office complex in order to estimate the flexibility potential of EVs. Thof low voltse consisting of real transaction data for 42 EVs charging for over a year at Utrecht Science Park, Utrecht, the Netherlands. The simulations showed that in 2050, 4 out of 7 studied transformers would be overloaded. Further, they concluded that around 50% of the EV demand can be delayed for more than 8 h. When this flexibility is used, overloading of 3 out of 4 transformers could be mitigated.

In ref. [103], authors analyzed how EV integration with the grid would cause overloading of the low voltage networks. A number of low voltage networks were modeled using DigSilent Powerfactory, taking into consideration the s variability of household electricity consumption, EV usage, and solar irradiance. Simulations showed that in urban networks, EV integration would cause higher cable loading, while in rural networks, it would cause voltage drops.

In ref. [104], authors provided a comparative analysis of how the common DC and AC bus architectures of fast chargers would impact the power quality of the grid.

In ref. [105], authors compared different charging strategies and ranked smart charging highly to tackle the adverse impact of EV charging on the grid.

In ref. [106], authors proposed a novel approach for charger placement considering cost, accessibility index, and VRP index. VRP index has the capability to consider the impact of EV charger on voltage stability, reliability, and power loss under a single framework.

In ref. [107], authors modeled the impact of EV chargers on a Latin American grid and concluded that with 10% EV penetration, the operational parameters of the grid would be least affected.

The impact of increased EV adoption on the Norwegian electrical grid was examined by the authors of ref. [108]. Calculations revealed that at a 20% EV penetration level, the system's weakest power cable would be overloaded. There was a 50% EV penetration into the network without any noticeable changes in voltage at any of the end-users.

As an example, in ref. [109], the authors simulate the charging infrastructure placement problem in the context of Guwahati city, India, and discuss the effect of charger placement on the voltage stability, dependability, and power losses of the city's streamlined power network.

Authors in ref. [110] used a stochastic bottom-up method to examine the impact of EV charging on load profiles. Filtering by demographic and economic characteristics, they examined a massive dataset of German mobility consisting of 70,000 car trips. It was determined that, depending on the loading infrastructure, the peak load may increase by a factor of 8.5. Meanwhile, peak demand would grow by a factor of around three, and yearly electricity consumption would roughly quadruple.

In ref. [111], the authors examined the effects of EV charging demand on a typical IEEE 33 bus distribution network across six different scenarios involving the placement of EV chargers. It was found that rapid charging stations may be placed at the strong buses up to a specific level without negatively impacting the operation of the power grid, but that placing such stations at the weak buses would have a negative effect.

The voltage profile of an IEEE 33 bus distribution network was investigated in [112] to determine the effects of a charging load for electric vehicles. The load voltage deviation

(0.062), total active power loss (120 kW), and total reactive power loss (80 kVar) were all influenced by even the smallest increment in EV charging load.

According to the findings presented in [113], various charging procedures can have a significant effect on a power grid. The benefits of smart charging solutions are discussed, including their potential to reduce voltage violations and cut down on expenses.

The authors of [114] suggested a multi-agent system to model the behavior of electric vehicles (EVs) charging at an energy hub where the penetration rate and charging pattern of the EVs vary. Vehicle-to-grid (V2G) connectivity, rapid charging (RCP), and smart charging (SCP) were all simulated. At 20% increments, the percentage of households using electric vehicles rose from 10% to 90%. When compared with a scenario without EVs, peak demand increases by 3.4% to 17.1% under the UCP. The SCP moves the EV charging load to the valley period, so the EH's energy dispatch between 07:00 and 23:00 is unchanged from the reference scenario. The highest grid electricity demand occurs when V2G is considered; for instance, the demand with 50% PR is twice the grid electricity demand in the reference situation.

Researchers in [115] examined the effects of adding EV chargers to the load of the IEEE 33 bus test network using standard reliability indices such as SAIFI, SAIDI, and CAIDI. In the baseline scenario, SAIFI was estimated to be 0.0982 interruptions per year. The SAIFI went up to 0.1080 interruptions per year when a rapid charging station with 30 servers was installed on bus 11, the system's strongest bus. In addition, the SAIDI and CAIDI values rose to 0.5374 and 5.2481 h per year and interruption, respectively.

8. Impact of RES Integration to Grid

Table 4 summarizes the results of RES integration to the grid.

Ref	Author	Year	Parameter	Diligence	
[116]	Wang et al.	2022	energy bases and load centers	Enhances power generation of renewable energy source	
[117]	Shiwei Yu	2022	Direct current (DC) and alternating current (AC) transmission technologies	Increased power transfer capability	
[118]	Bhattacharya et al.	2021	marine renewable energy	Marine renewable energy resources are more available and persistent	
[119]	Akhtar et al.	2021	Fault analysis	Risk assessment of restructured power system with integrating the electric vehicles and renewable energy resources	
[120]	Hanni Wirawan et al.	2021	Rural electrification	An off-grid renewable energy-based electricity rollout to alleviate rural poverty in Indonesia's remote islands	
[121]	Mark et al.	2021	Energy	Stable grid	
[122]	Mokeke, Leboli Z et al.	2021	Voltage, frequency and rotor angle	increased penetration of the intermittent (variable) renewable energy generators (IREGs) led to grid instability	
[123]	Ayadi et al.	2020	Smart grid	Efficient Energy management	
[124]	Oyekale et. al	2020	Renewable hydrogen	Cheaper energy storage devices	
[125]	Alam et al.	2020	Energy storage devices, and fault current limiters	high-level RES grid integration issues	
[126]	Poul Alberg et.al	2020	Energy systems	Energy system integration, effects, and environmental performance are analyzed.	

Table 4. Research work on impact of RES on power grid.

Ref	Author	Year	Parameter	Diligence
[127]	G. Sokhna Seck	2020	Energy	Power exchanges with neighbors with higher share of renewable energy systems in the power production
[128]	Ali et al.	2020	Energy	Improved stability
[9]	Johnson et al.	2020	Electricity generation	Improved stability
[129]	Samuel C. Johnson	2020	Grid operation	non-synchronous energy generation
[130]	Keck, Felix	2019	Energy	Stability Improved
[130]	Worighi et al.	2019	Energy storage system	Improved installed capacity
[131]	Ndamulelo et al.	2019	Grid-connected variable renewable energy	Examines generator type, penetration, and grid features of international VRE integration
[30]	Samuel et al.	2019	Energy	Improved grid stability
[132]	Madeleine McPherson et al.	2018	Energy	Improved stability
[133]	Perera et al.	2017	Energy	Improve grid stability
[134]	Musau et al.	2017	Energy	This reduces generation costs and pollution
[135]	Triviño et al.	2017	Smart grid	Voltage improvement
[136]	Harrouz et al.	2017	Energy	Grid voltage improvement
[137]	Schmietendorf et al.	2017	Energy	Increased grid reliability

Table 4. Cont.

Transition regression panel smoothing reflects the regional effects of grid-connecting renewable energy sources, as studied by the authors of ref. [116]. In energy bases and load centers, higher voltages generally have a more beneficial effect on renewable energy output and consumption, but this is not always the case. Ultra-high voltage systems are unaffected by the incorporation of renewable energy sources into the grid. No renewable energy source is currently feasible due to the lack of extra-high voltage power lines serving as the backbone of the national grid. The power grid needs to be modernized immediately.

In ref. [117], we determine the optimal line path, transmission capacity, and development schedule for six inter-regional transmission lines. By 2039, transmission from the northwest to the east will increase by 265%, while transmission from the north to the center will increase by 160%. The current standard of 400 kV DC (5 GW) will be replaced by 800 kV DC (10 GW) in 2033. The highest construction years are 2036–2039. Renewable wind and solar power are generated in central and eastern China. Increases in wind and solar electricity are expected to treble by the year 2039. By disconnecting lines 2–6 and 7–9, energy storage and demand-side response can increase renewable power on the grid by 1.7% and 2.6%, respectively.

Based on their findings, the authors of ref. [118] concluded that marine renewable energy resources are consistently more available and persistent than wind and solar on an hourly basis during the whole year of operation. There is also speculation that using wave resources can reduce the amount of balancing effort required by electrical grids.

The method presented in [119] combines the encouraging effects of a Gaussian distribution with the probabilistic breakdown of many components' failure rates inside a fuzzy fault tree framework. System switches and low power components were not detectable by traditional fault tree analysis. The possibility of power disruptions is also ignored. Lack of data causes significant failure and poor behavior forecast uncertainty for grid-connected wind energy power systems and EV installations.

In order to determine if rural electrification through renewable energy village grids (RVGs) may alleviate poverty in off-grid villages and islands in Indonesia, the authors of [120] conducted an in-depth study. This research looks at how the use of renewable off-grid electricity could help alleviate poverty and security concerns in far-flung areas. Energy production is constrained by geography. DID compares the outcomes of treatment and nontreatment in 217 remote villages in Indonesia. Ninety-one people from marginal socioeconomic backgrounds were wiped out by the program. This research also found that providing small businesses in the hamlet with access to electricity helped reduce poverty. The use of renewable energy to power homes off the grid has helped alleviate poverty on Indonesian islands.

Using a scenario in which every country in Western Europe relies only on renewable sources of energy such as wind, water, and sunlight, the authors of [121] examine the effects of interconnecting vs. isolating their electric networks on energy costs and demand (WWS). Wind, sun, thermal loads, and refrigeration loads can all be predicted with the use of weather models. World Wide Solar Power, Storage, Demand Response, Power, Heat, Cold, and Hydrogen are all balanced by grids. The United Kingdom, France, Germany, Spain, Italy, and Spain all have dependable options, as do Luxembourg and Gibraltar. Energy pricing, generator/storage overbuilding, energy shedding, and land/water demands can all be reduced through interconnected nations. Electrical costs in Western Europe might be reduced by up to 13% if the region were linked. The most significant reduction in emissions occurs when Denmark (20.6%) and Northwestern Europe (13.7%) are connected to Norway's abundant hydropower. Connections between Luxembourg and larger states are advantageous for everyone. Countries such as France and Germany, among others, have the financial means to switch to 100% WWS grids.

Utility-scale PV and wind farm integration is being driven by the need for sustainable power systems. Wind and solar photovoltaic (PV) systems are distinct from traditional power plants in that they generate their own electricity [122]. Lesotho's electricity supply is unstable due to intermittent renewable energy production (IREGs). The majority of PV and wind generators were located at the Ha-Ramarothole and Letseng substations. Research on the dynamic effects of changes in renewable energy capacity used a short circuit defect at the bus bar with the shortest critical clearance time (CCT) to quantify changes in voltage, frequency, and rotor angle. Study of steady-state voltage with hourly load data, IREG data, and Muela Hydropower generation for 2018. The stability of voltage, frequency, and rotor angle was measured by the Lesotho Grid Code.

With the current energy problem, renewable energy sources are expected to replace traditional power plants within the next several decades, as discussed in [123]. Therefore, the focus of the present research is on finding ways to integrate renewable energy sources into the smart grid. This research explains the advantages and disadvantages of using various control systems, all of which have played a role in the efficient incorporation of renewable energy sources.

The consequences of renewable energy technology grid integration on power network efficiency and the most prevalent approaches to addressing these issues are summarized in [124]. Renewable energy can be obtained from the sun, the wind, biomass, geothermal heat, and renewable hydrogen/fuel cells. Many global energy projections have incorrectly stated that renewable resources can provide global energy needs in the thousands. This is because of limitations in the actual use of these resources. For a comprehensive summary of these challenges and workable answers, this review research is essential.

In the last two decades, RESs have become increasingly commonplace [125]. Without regular synchronous generators, the system has less inertia, making regulation more difficult. High uncertainties, low fault ride through capability, high fault current, limited generation reserve, and poor power quality are just some of the technical difficulties that arise from RESs integration. Solar and wind power are risky because of the unpredictability of the sun and the wind. In order to address these problems, cutting-edge technologies have been developed for control, optimization, energy storage, and limiting fault currents.

The report analyzes the systemic challenges associated with integrating RES into grid infrastructure. Answers to the challenges are being discussed. The problems with and potential solutions for combining wind and solar energy are extensively documented. Lastly, some considerations for renewable energy integration are given for experts in the field and academics.

With an eye toward technology, availability assessment, and system integration, the authors of ref. [126] examine the current state of renewable energy studies. Renewable energy sources such as wind, waves, geothermal heat, solar power, and electricity, and salinity gradient systems are discussed. Environmental performance, environmental consequences, and the integration of energy systems are evaluated in the last section. This review provides a broader context for the research presented at the Sustainable Development of Energy, Water, and Environmental Systems (SDEWES) conference series and published in Special Issues of many journals.

The writer of ref. [117] examined the current state of the renewable energy study, concentrating on technological advancements, availability evaluations, and system integration. Technology using wind, waves, geothermal heat, solar power, and electricity, and electricity generated by a salinity gradient are discussed. Integrating energy systems, their consequences, and their impact on the environment are evaluated at the end.

The short-term performance of the French electrical grid is the subject of a long-term prospective study in France. After Re-union Island, countries that rely heavily on electricity look at their infrastructure and evaluate its adequacy and temporary stability. The TIMES-FR Energy System Optimization Model uses kinetic reserves as a reliability metric for the French power sector (ESOM). Modular renewable energy backup stabilizes power systems. Consistency is guaranteed at 65% VRE. The highest hourly VRE for 100 EnR reliable power generation was 84%. Completely relying on renewable energy sources (RES) would triple new installed capacity between 2013 and 2050, significantly improving the reliability of the system. To meet the dependability need at any time by providing the system with greater inertia, it is important to think about thorough upstream planning and flexible solutions such as demand-response, storage technologies, linkages, or replacement or new facilities. Emphasize electricity trades with neighbors who are producing renewable energy [118].

Integration and renewables, as the authors of ref. [119] showed, pose a threat to the reliability of the electricity grid. Grid connectivity was disrupted when wind and solar PV replaced traditional power plants. Grids are made more reliable by modern technology. The integration of renewable power plants and regulations for their use are reviewed here. Requirements for grid stability are compared, including voltage stability, frequency stability, voltage ride-through (VRT), power quality, and active and reactive power regulations. Controls. In this research, cutting-edge methods of regulation and control are weighed and compared. Overall, the study finds that integrating requirements enhances grid operation, stability, security, and reliability; however, protective rules, global harmonization, and control optimization all need improvement. Limitations on the use of RES. Developers and researchers could benefit from this review. Assist power grid operators worldwide in creating uniform electrical standards.

According to ref. [120], current energy needs far exceed those met by more traditional means. The use of electricity is crucial to technological advances. Most pollution associated with energy generation comes from burning fossil fuels. Electricity supply and demand mismatches can be closed by renewable energy. A decrease in carbon dioxide emissions (GHG). Energy production is based on location. Power quality, reliability, stability, harmonics, single-phase oscillations, and reactive power adjustment could all be impacted by the incorporation of RES into the grid. The problems with RES interruptions are eliminated when an ESS is integrated. Ecology-friendly power Increasing the reliability, effectiveness, and energy density of renewable energy generation systems is the focus of RES-ESS innovation.

Traditional energy sources are insufficient to supply the world's energy demands [121]. Electricity is the lifeblood of today's manufacturing and scientific communities. The vast majority of us rely on fossil fuels for our energy needs. The gap between electricity supply

and demand could be closed by renewable energy. Carbon Dioxide Produced by the Energy Sector (GHG). Energy generation is based on consumption. Reactive power correction, RES integration, and power quality/reliability/stability/harmonics/single-phase overcurrent could be negatively impacted. The underutilized RES and ESSs help out. absolutely safe for use by anyone. ESS/RES. The power density, efficiency, and dependability of RES power systems are all improved by ESSs. The effectiveness of PV systems is diminished by harmonic overtones. Control strategies using multiple FACTS types have an impact on RES-based power grids. In order to integrate RES into electricity networks in a secure manner, FACTS are required.

Given the flexibility EES provides the power network, it may one day be possible to make the switch to clean energy, as demonstrated by [122]. Many professionals acknowledge EES's value, but many express concern about its unpredictability in key areas such as technology, price, business strategies, and market architectures. Cost-benefit studies of EES for zero-emissions electricity generation are performed here. The effects of adding EES to wind and solar generation on LCOE, installed capacity, generation mix, and energy spillage are investigated in a GIS-supported hourly simulation study of Australia. There is a reduction in LCOE when EES is used in settings with high penetration of renewable energy sources. Costs of less than one thousand Australian dollars per megawatt-hour make 90–180 GWh of EES a practical option in Australia. In addition to reducing LCOE by 13–22%, 22–23% of installed capacity, and 76% of energy loss, the study finds that EES can improve efficiency. The generation mix is profoundly impacted by EES deployment.

Variable renewable energy (VRE) systems that are connected to the grid have been growing rapidly in recent years [123]. The addition of more of these devices complicates plans to upgrade regional power grid infrastructure. The high VRE regional power networks were analyzed. Renewable resources, VRE goals, and grids vary by region. The study of VRE integration is primarily conducted on a regional scale. Since it would be impractical and prohibitively expensive to do comprehensive VRE integration studies for every grid, it is essential to narrow the scope by identifying anticipated regional difficulties. This research looks at the many types of generators being used around the world, as well as their penetration rates and how they interact with the power system. The integration of VREs offers regional advantages.

It is clear from ref. [124] that renewable energy sources such as wind and solar threaten grid reliability. The purpose of this study was to determine under what conditions, in terms of frequency contingencies such as generator outages, an electric system would be more vulnerable in the presence of a high penetration of renewable energy sources. By proxy, system inertia was evaluated using unit commitment and dispatch modeling, and grid stability was measured. A case study of Texas' grid showed how this could be done. Modeled scenarios showed that the Texas grid can adapt to significant changes, even with a high penetration of renewable energy (30% of energy generation, up from 18% in 2017). Without the addition of nuclear power plants and exclusive-use networks, the model exhibits unstable inertia. To preserve system inertia, our model ran a large number of coal and natural gas combined-cycle facilities at part-load or the lowest operational level possible. If the share of renewable energy increases, this could affect other electrical grids that rely on synchronous generators for inertial support.

A rise in variable renewable wind and solar resources [125] necessitates a rise in demand response, dispatchable power, transmission connectivity, and storage. In this piece, we look at how to measure the value of storage facilities in terms of their production costs. There is still a need for storage for non-renewable energy sources since they are so inflexible, even at low levels of renewable energy penetration. For high penetrations of renewable energy, storage is essential for flexibility. The rate at which storage assets are used is affected by factors such as the proportion of renewable energy sources in the energy system, the format of the bidding process, and the ownership model. Consumption of storage assets will occur at cheaper times and production at more costly times regardless of the bidding structure, but the largest price differential can be achieved by the central authority

bargaining for price arbitrage. The value of storage is based on system adaptability, renewable energy share, and competitive bidding.

Traditional generators and rotational systems are being phased out in ref. [126] in favor of grid-connected wind and solar PV. The price of electricity and pollutants is lowered as a result. The frequency dynamics are accelerated by low inertia, which might be detrimental to stability. Both frequency regulation and stability are hampered as a result. Complete blackouts can be caused by destructive vibrations (DV) and under frequency load shedding (UFLS) if the frequency variation is particularly large. Grid expansion in Kenya was aided by renewable energy sources such as wind and solar. When a monkey accidentally tripped a transformer at the Gitaru Hydroelectric Power Plant at 11:30 a.m. on Tuesday, 7 June 2016, the entire country lost power. In order to better understand frequency instability with RE, this work revisits UFLS and proposes the concept of Combined Fre-quency with Renewable Energy Storage Cost (CFS). Recaps the Kenyan Court Scandal.

Renewable energy sources (RES) [127] are crucial to the operation of smart grids. The effect that renewable energy has on the electrical grid is determined by the nature of the source, its penetration rate, and the design of the grid. In this study, we analyze the effect of renewable energy sources on a measurement of power quality (voltage dips). Consider the Italian grid and its structure. Considering how renewable energy sources and grid connectivity can change from region to region, this concept is intriguing. In this study, we test the hypothesis that the Pearson's index and the *p*-value have a linear relationship. Less voltage drops mean more renewable energy sources. We disprove this assumption.

Issues with increased demand for electricity from the grid are described in ref. [128]. Standardized resources can't keep up with consumer need. It's clear that, in this case, renewable energy is the most economical and beneficial option for satisfying household power demands. The quality and security of renewable energy systems are enhanced through the use of information and communication technology in a smart grid. The potential of renewable energy sources in Algeria is analyzed.

Changes in renewable energy inputs threaten grid reliability, as stated in ref. [9]. Inefficient load balancing and self-organized synchronization lead to short-term oscillations in such systems. During these times, electricity generated by wind and solar is unpredictable and not Gaussian. Using Kuramoto's power grid model, we investigate the effects of short-term wind changes on desynchronization, frequency, and voltage quality. Changes in the feed-in are represented by a temporal correlation, a Kolmogorov power spectrum, and intermittent increases. We discovered that correlations are required to capture the probability of severe outages, but the intermittent nature of wind power has significant effects on power quality, as the intermittency is directly transferred into frequency and voltage fluctuations, leading to a novel type of fluctuations that is beyond engineering knowledge.

9. Challenges of EV Integration to Grid

The safe operation of the power network may be compromised with the increasing EV numbers. EVs have the capacity to serve as additional load as well as distributed flexible load. The challenges imposed by the grid integration of EVs can be divided into power issues, energy issues, and grid reinforcement. A summary of these challenges is presented in Table 5.

Use Case	Locations	Power Issues	Energy Issues	Grid Reinforcement
Public slow charging	Street parking, Social/recreational areas, Park & Ride	For multiple installations, significant impacts can be expected in Secondary Substations (MV/LV transformers)	No significant issue	It could be necessary to replace MV/LV transformers and/or MV and LV feeders

Table 5. Challenges of EV integration to grid [138].

26 of 39

Use Case	Locations	Power Issues	Energy Issues	Grid Reinforcement
Home private charging	Single houses, apartments, hotels, offices	Voltage issues can be expected in rural areas	No significant issue	It could be necessary to replace MV/LV transformers and/or MV and LV feeders
High Power Chargers—"Fuel Station" Model	Fast chargers (50–150 kW) in existing fuel stations	Single installations may require a significant increase of power absorption	Energy withdrawal from the network could be significant	It could be necessary to install a dedicated MV substation with additional cost and time. MV lines (and in some cases, MV/LV transformers) could need to be replaced
Urban hyper hubs	Hyper fast chargers (150–350 kW) in new dedicated areas designed for cars in urban areas	Loads generated by EV charging add up to other LV and MV loads. The impacts could be significant, also on MV lines	No significant issue	MV lines (and in some cases, MV/LV transformers) could need to be replaced
Bus depots	High number (tens/hundreds) of buses performing night charging	A single deposit could require 5–10 MW, often in urban areas. There is a strong need for coordination between grid operators and local public transport operators	Moderate additional energy demand	In the event of the high number of buses, new primary substations could be required
Highway hyper hubs	Hyper fast chargers (150–350 kW) in new dedicated areas on highways both for cars and for heavy duty vehicles	A single hub could require more than 10 MW, often in rural areas. There is a strong need for coordination with grid operators in order to locate hubs close to existing HV lines	Energy withdrawal from the network could be significant but no issues are expected	A new Primary Substation would be required. A well-planned location would minimise the need for new HV lines
Company fleets	Pool vehicles (utilities, public services, private companies)	For multiple installations, significant impacts can be expected in Secondary Substations (MV/LV transformers)	No significant issue	It could be necessary to replace MV/LV transformers and/or MV and LV feeders

Table 5. Cont.

10. Challenges of RES Integration to Grid

Integrating RES to the grid may have several challenges such as flicker, harmonic distortions, narrow voltage trips as listed in Table 6.

Table 6. Challenges of RES integration to grid [139].

Category	Challenges
Quality	Increasing flicker Increasing harmonic distortions Unreliable shut down during blackouts Increasing local voltage excursions
Flow	Missing distribution grid capacity Increasing volatile flow patterns from lower grid levels Inadequate protection design Increasing short circuit current Narrow voltage trip details Missing transmission grid capacity

Table 6. Cont.

Category	Challenges	
Stability	Insufficient reactive power provision Decreasing level of short circuit power Decreasing level of inertia Inadequate coordination of frequency and voltage trips limit	
Balance	Insufficient short term generation adequacy Insufficient forecasting of variable RES Insufficient firmness of variable RES generators	

11. Optimization Techniques for EV and RES Integration to Grid

11.1. Optimization Techniques for EV Integration to Grid

The research works on applications of optimization techniques for integrating EV to the grid are elaborated here.

Zhang et al. proposed a wind energy reduction usage strategy that utilizes the use of electric vehicles (EVs) with the aim to minimize the variations in wind power output by utilizing a hybrid algorithm called Implementation of Differential Evolution Quantum Particle Swarm Optimization (IDE-QPSO). The technique starts with a charging load demand for EVs. The second step is to use an enhanced particle swarm optimization algorithm to explain a multi-objective optimization framework that takes into account the utilization of wind energy shortfalls while also minimizing the intensity and variability of wind energy output. S sufficient modelling trials demonstrated that this method may successfully mitigate the impact of wind power production variability while simultaneously increasing the efficiency with which wind power is used [140].

Huy et al. introduced a comprehensive Home Energy Management System (HEMS) framework that incorporates Electric Vehicles (EVs) and employs a Multi-Objective Mixed-Integer Linear Programming model. The model is designed to optimize various parameters, such as energy costs, peak-to-average ratio (PAR), and discomfort index, while also maximizing vehicle-to-home and home-to-grid capacities (DI)In addition, a strategy for integrating the enhanced-constraint to efficiently tackle any multi-objective HEMS issues is described. Multiple computations employing both deterministic and stochastic scenarios are used to verify the suggested method [141].

Li et al. presented in their research a layered control system for plug-in hybrid electric vehicles to reduce their fuel usage. In the optimization method, two kinds of powertrain concepts are utilized initially. Using convex optimization, an approximation of a model is applied in the higher control layer to obtain the optimum baseline state of the charge trajectory. Consequently, at the deeper control plane, the fuel usage is reduced in real-time utilizing a high-fidelity powertrain framework and integrating the corresponding consumption minimize method into the adaptive control scheme. Furthermore, optimization findings from the remaining three real-time control techniques and two prospective energy management techniques are provided and examined to validate the suggested method's efficacy [142].

Wang et al. examined EV and proposed an effective co-optimization technique in order to concurrently identify optimum battery pack and turbine-generator pack sizes and appropriate control parameters. To design an objective function, the starting mass, fuel usage, and battery deterioration are selected as optimization targets. Then, a new enhanced hypotrochoid spiral optimization method (EHSOA) is developed to solve the complex co-optimization issue. An upgraded mechanism is initially introduced in this approach to prevent the optimization method from being caught in local optima. Compared with the original design, the suggested technique decreases initial mass by 5.08 percent, fuel usage by 26.10 percent, and battery deterioration by 2.0 percent. Eventually, it is proved that the suggested EHSOA is superior to existing optimization methods for solving the difficult co-optimization issue [143].

Yang et al. demonstrated a complete multi-objective optimization strategy for energy planning in local multi-energy frameworks with plug-in electric vehicles (PEVs), with the goal of enhancing the revenue of the companies in the local multi-energy frameworks while minimizing their CO₂ emissions. This subject concerns information exchange in local multi-energy infrastructure and determining optimal charge/discharge techniques for plug-in electric vehicles (PEVs) in order to optimize the profit while simultaneously reducing CO₂ emissions. It may be addressed by establishing a multi-purpose goal and a multi-objective programming problem by accurately describing the interdependencies involving power generation. Depending on the framework, a Modified Group Search Optimization (MGSO) method is employed to tackle this issue. The structural model dramatically improves local and global search. According to the outcomes of the potency test of the optimization method for maximizing the manufacturer's revenue while simultaneously lowering CO₂ emissions, this goal is attainable through optimized coordination of multiple power generation in local multi-energy structures and efficient implementation of the customization gathered on both the demand and supply sides [144].

Vasanth et al. have developed an enhanced version of the Wild Horse Optimizer called the Improved Wild Horse Optimizer with Deep Learning-enabled system for Battery Management (IWHODL-BMS). This approach is specifically designed for Internet of Things (IoT) based Hybrid Electric Vehicles (HEVs) and focuses on battery management. The IWHO method is used as a hyperparameter optimizer to improve the SOC estimate efficiency of the ABiGRU method. The implementation of the ABiGRU model leads to an easier and more precise portrayal of the input. A modeling and simulation result revealed the superior performance of the IWHODL-BMS model in comparison to alternative approaches, as measured by a variety of parameters [145].

11.2. Optimization Techniques for RES Integration to Grid

The research works on applications of optimization techniques for integrating RES to the grid are elaborated here.

Nayak et al. formulated an inherited competitive swarm optimization (ICSO) algorithm for the optimal placement and rescheduling of the battery energy storage systems in combination with the wind energy system. The formulation of the ICSO has been done based on human generation aging, in which the characteristics of the elder most on the generation is transmitted to their offspring. The algorithm proved to be better than the grasshopper algorithm [146].

Wang et al. developed a modified coot optimization algorithm (MCOA) for the optimal generation of the wind energy considering the various climatic changes. The MCOA has been based on the self-adaptive weighing technique which assist it to prevent from trapping in the region of the local optimal solutions. The penetration of the wind energy system and its operating cost has been significantly reduced with the application of the MCOA [147].

Xian and Che implemented Whale Optimization Algorithm (MWOA) with the association of support vector regression for the forecasting of the wind speed. They utilized the WOA to determine the optimal site and location of the wind energy system that would benefit the generation of power depending upon the intermittence of the wind speed [148].

Ullah et al. in their research developed a multi-objective wind driven optimization and multi-objective genetic algorithm to optimize the operation cost and the emission related to the power dispatch. The researchers also examined a combination of demand side management and incline block tariffs, which take the behavior of industrial and commercial customers into account [149].

Liu et al. analyzed the economic dispatch considering the thermal, wind, and solar energy system. In their work, they developed multi-objective moth flame optimization techniques to achieve the optimal performance of the economic dispatch model [150].

Rawa et al. developed an efficient optimization technique with the combination of the runga kutta and gradient optimizer to solve the transmission expansion planning. The planning for transmission was created by incorporating a wind and solar energy system, along with a battery energy storage system [151].

Krishnan and Kumar formulated a hybrid optimization approach based on the combination of Gurra fish optimization and isolation forest optimization for computing the optimal performance of the controller parameters for a microgrid. The complete microgrid framework is comprised of a solar photovoltaic system and a renewable energy system. The results achieved with the optimization technique have been superior in comparison to the optimization algorithms such as artificial bee colony optimization and ant lion optimization techniques [152].

Jiang et al. developed a demand response considering the economic operation of the microgrid with the influence of renewable energy sources. They utilized an improved wave optimization algorithm for their problem. The model WWO behavior is influenced by water waves. Waves prevailing on the ocean's surface contain complicated yet intriguing interactions that may be exploited to address optimization issues. In this method, every solution to a problem is symbolized as a water wave, which undergoes modifications in the search space as a result of three actions: propagation, refraction, and decay. This method is preferred because of its straightforward structure, elitism, and the ability to escape from local optima due to the presence of multiple search agents and the recombination of generations. Thermal and electrical dynamic load sharing and a thermal energy storage system led to a reduction in operating expenses, according to the findings [153].

11.3. Summary of Optimization Techniques for EV and RES Integration to Grid

The summarization and the research output ached with the various optimization techniques applied in the field of EV and RES by the research, which can be found in Table 7.

Ref No.	Year	Торіс	Implemented Optimization Techniques	Research Description
[140]	2022	EV	Differential Evolution Quantum Particle Swarm Optimization Algorithm (IDE-QPSO)	Taking inspiration from the Differential Evolution Algorithm, they have introduced the differential evolution operator to the iterative process of particle evolution, resulting in the creation of the IDE-QPSO algorithm
[141]	2022	EV	Augmented ε-constraint method and lexicographic optimization	A technique for effectively dealing with multi-objective HEMS (Home Energy Management Systems) problems is introduced, which involves combining the augmented ε-constraint method with lexicographic optimization
[142]	2022	EV	Convex optimization	Optimizing fuel economy in a PHEV involves a hybrid algorithm that integrates Convex Programming (CP) with real-time energy management strategies (EMSs) resulting in a CP-ECMS predictive EMS
[143]	2022	EV	Enhanced Hypotrochoid Spiral Optimization algorithm (EHSOA)	The first step in this algorithm involves proposing an improved bi-considering mechanism that prevents the optimization process from getting stuck in local optima
[144]	2022	EV	Modified Group Search Optimization (MGSO)	The suggested framework leads to a substantial enhancement in both local and global search
[145]	2022	EV	Improved Wild Horse Optimizer (IWHO)	The hyperparameter optimization process makes use of the IWHO algorithm

Table 7. Summarization and contribution of optimization techniques applied in the field of EV and RES.

Ref No.	Year	Торіс	Implemented Optimization Techniques	Research Description
[146]	2022	RES	Inherited Competitive Swarm Optimization (ICSO) Algorithm	Enhanced the coordination between the exploration and exploitation
[147]	2022	RES	Modified Coot Optimization Algorithm (MCOA)	The self-adaptive weighting technique and turbulent technique are employed in this method to avoid getting stuck in local optimization, ultimately enhancing the accuracy of predictions beyond what is achievable with the original Coot Optimization Algorithm (COA). The modified COA metaheuristic model, which employs these two techniques, displays a lower standard deviation when compared to other metaheuristic algorithms
[148]	2022	RES	Whale Optimization Algorithm (WOA)	Unified optimization and the WOA are used to solve the parameter selection problem. The global selection of parameters is achieved by the unified optimization, which considers the interactions between different models
[149]	2022	RES	Multi-Objective Wind-Driven Optimization (MOWDO) Algorithm and Multi-Objective Genetic Algorithm (MOGA)	Compared to existing models that involve or do not involve hybrid schemes, the energy optimization model improves the performance of a smart microgrid by minimizing its operational cost, lowering pollution emissions, and enhancing availability
[150]	2022	RES	Moth-Flame Optimization Algorithm (MFOA)	To address the high-dimensional, non-linear, and non-convex nature of the HDEED (High-Dimensional Expensive Optimization Problem with Expensive Constraints), a new optimization algorithm called MFO_PDU (Moth-Flame Optimization algorithm with Position Disturbance Updating strategy) has been introduced
[151]	2022	RES	Hybrid Runge Kutta Optimizer-Gradient-Based Optimizer (HRKOGBO)	Improved solution and avoidance of local optima.
[152]	2021	RES	Hybrid Garra Rufa Fish Optimization (GRFO)—Isolation Forest (Iforest)	The GRFO-iForest technique surpasses other methods in both steady-state and transient operation. It enhances the sustainability of the microgrid by restoring it to its normal operational status after minor physical disruptions and maintains stability by utilizing regulators in an optimized manner
[153]	2021	RES	Improved Water Wave Optimization Algorithm (IWWOA)	In this algorithm, each solution to a problem is depicted as a water wave, which undergoes modifications in the search space due to three behaviors: propagation, refraction, and decay of the waves or problem solutions

 Table 7. Cont.

The integration of electric vehicles (EVs) into power systems is an active area of research, and there are several directions that researchers are currently pursuing to optimize the integration. Some of the research directions are:

• Optimal charging and discharging strategies: Researchers are investigating optimal charging and discharging strategies for EVs to minimize the cost of electricity and reduce the impact of EVs on the power grid. Optimization techniques, such as dy-

namic programming and model predictive control, are being used to find the optimal charging and discharging schedules.

- Vehicle-to-Grid (V2G) technology: V2G technology allows EVs to discharge their stored energy back into the grid, which can help balance the power supply and demand. Researchers are exploring the optimal scheduling of V2G systems to maximize the benefits to both the EV owners and the power system operators.
- Energy management systems: Energy management systems (EMSs) can control and optimize the charging and discharging of EVs in real-time based on the power grid's current state. Researchers are exploring the development of EMSs that can take into account factors such as the availability of renewable energy sources, grid congestion, and user preferences.
- Battery degradation: Battery degradation is a critical issue for EVs. Researchers are exploring optimization techniques to extend the life of EV batteries by minimizing battery usage, optimizing charging and discharging schedules, and managing battery temperature.
- Grid stability and security: The integration of EVs into the power grid can impact grid stability and security. Researchers are exploring optimization techniques to ensure that the power grid remains stable and secure with the increasing penetration of EVs.

The application of optimization techniques in renewable energy system integration in power system grid is a rapidly evolving research field. Some current research directions in this field are:

- Optimal power flow (OPF): OPF is an important optimization technique used to optimize the operation of power systems. Recent research has focused on developing new algorithms to solve the OPF problem in power systems with high levels of renewable energy penetration.
- Energy storage optimization: Energy storage systems (ESS) are increasingly being used in power systems to improve the integration of renewable energy sources. Research in this area is focused on developing algorithms to optimize the use of ESS to minimize system costs while ensuring the reliability of the power system.
- Demand response: Demand response (DR) programs allow consumers to adjust their energy consumption in response to changes in electricity prices or grid conditions. Research in this area is focused on developing algorithms to optimize the use of DR to minimize system costs and reduce the need for additional capacity.
- Renewable energy forecasting: Accurate forecasting of renewable energy production is critical for the effective integration of renewable energy into power systems. Research in this area is focused on developing improved forecasting models that can account for the variability and uncertainty of renewable energy sources.
- Distributed generation: Distributed generation (DG) refers to the generation of electricity from small-scale sources located close to consumers. Research in this area is focused on developing algorithms to optimize the use of DG to minimize system costs while ensuring the reliability of the power system.

Overall, the application of optimization techniques in EV integration is a promising area of research that can help to maximize the benefits of EVs while minimizing their impact on the power system grid. In addition to this, the application of optimization techniques in renewable energy system integration in power system grid is a dynamic and rapidly evolving research field, with a focus on improving the efficiency, reliability, and sustainability of power systems with high levels of renewable energy penetration.

12. Discussions

This review puts forward the latest research developments in the paradigm of EV and RES integration into the grid. A total of 153 documents, including reports, journal articles, and conference papers, are meticulously reviewed in this work. Quantitative analysis of EV and RES integration into power system grids typically involves modeling the power system and simulating different scenarios with varying levels of EV and RES penetration.

The integration of EVs and RES presents both challenges and opportunities for the power system grid. Some of the key parameters that are typically considered in the reported works include:

- 1. EV and RES penetration rates: Higher penetration rates can reduce greenhouse gas emissions and improve air quality, but they can also present challenges for the power system grid in terms of grid stability and reliability.
- 2. Charging patterns and RES generation variability: The timing and duration of EV charging can have a significant impact on the power system grid, and the variability of RES generation can also impact the ability of the grid to balance supply and demand. Managing the charging patterns of EVs and the variability of RES is key to ensuring grid stability and reliability.
- 3. Grid capacity and flexibility: The ability of the power system grid to handle increased demand from EV charging and renewable energy generation depends on the capacity of the grid as well as its flexibility to adapt to changing demand and generation patterns. This can include the availability of energy storage systems and the ability to manage variable generation from sources such as wind and solar.
- 4. Transmission and distribution infrastructure: The integration of EVs and RES often requires upgrades to the transmission and distribution infrastructure, including new transmission lines and substations.
- 5. Energy storage: Batteries can serve as a medium of storage and help to smooth out the variability of RES and provide grid stability during times of high demand.
- 6. Economic and regulatory factors: The cost of EVs and renewable energy technologies, government policies and incentives, and market design can all impact the adoption of EVs and RES in the power system grid.

Using these parameters, quantitative analysis can help identify the potential impact of EVs and RES integration on the power system grid and inform policy decisions related to infrastructure upgrades, energy storage deployment, and regulatory frameworks for EV and renewable energy adoption. The analysis can also help identify the most cost-effective ways to integrate EVs and renewable energy sources into the grid while maintaining grid stability and reliability.

13. Conclusions

Global warming, pollution, and the depletion of fossil fuels have forced mankind to delve into alternate sources of energy and cleaner modes of transport. In recent years, RES has been massively introduced to the grid. Also, EVs are becoming popular as a cleaner mode of transport. However, the introduction of RESs and EVs to the grid has imposed additional challenges on the grid operators because of their random nature. Unplanned placement of EV chargers and uncoordinated charging may cause serious issues such as degradation of the voltage profile and reliability indices, harmonic distortions, and power losses. Similarly, increased RES penetration on the grid may result in power quality issues.

This review focuses on the integration of RES and EVs to the grid and reports comprehensively on the global status of RESs and EVs, impact of integrating RESs and EVs to the grid, challenges of integrating RES and EV to the grid and the mitigation techniques, soft computing applications to EV and RES. It has been observed that soft computing techniques are predominantly applied for solving different optimization problems related to RES and EV.

Our future work will address:

- Planning and operation of energy system having EV and RES
- EV integrated Virtual Power Plant (VPP)
- Metaheuristics for planning and operation of smart power networks

Author Contributions: Conceptualization, P.S. and S.D.; methodology, K.P.; formal analysis, S.S.; investigation, P.S.; resources, S.D.; data curation, S.S.; writing—original draft preparation, S.D.; writing—review and editing, S.D. and K.P. All authors have read and agreed to the published version of the manuscript.

Funding: This project has received funding from the European Union's Horizon 2020 research and innovation programme under the Marie Skłodowska-Curie grant agreement No. 945380.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

AC	Alternating Current
DC	Direct Current
BEV	Battery powered electric vehicle
EV	Electric Vehicle
HEV	Hybrid Electric Vehicle
RES	Renewable Energy Source
TCO	Total cost of ownership
RSP	Risk Sharing Program
MDP	Markov Decision Program
ToU	Time of Use
V2G	Vehicle to grid
THD	Total Harmonic distortion
SAIFI	System Average Interruption Frequency Index
SAIDI	System Average Interruption Duration Index
CAIDI	Customer Average Interruption Duration Index
AENS	Average Energy Not Served
CSO	Chicken Swarm Optimization
SOCP	Second Order Cone programming
VRP	Voltage Stability Reliability Power Loss
VPP	Virtual Power Plant
UCP	Uncontrolled charging pattern
RCP	rapid charging pattern
SCP	smart charging pattern
POI	Point of Interest
ССТ	Critical Clearance time
SDEWES	Water and Environmental Systems
VRT	voltage ride-through
GHG	Greenhouse gas
SSO	Single sign on
RES-ESS	Renewable Energy Sources-Energy Storage Systems
VRE	Variable Renewable energy
DV	Damaging Vibration
UFLS	under frequency load shedding
CFS	Energy Storage Cost
PAR	peak-to-average ratio
DI	Discomfort Index
HEMS	Home Energy management systems
EHSOA	hypotrochoid spiral optimization method
IWHODL-BMS	Improved Hild horse Optimizer with Deep learning-enabled
	system for battery management
ICSO	Inherited competitive swarm optimization
MCOA	Modified Coot Optimization Algorithm
MM-QR	Moment's Quantile Regression
MWOA	Modified Whale Optimization Algorithm

References

- 1. Renewable Energy—Powering a Safer Future United Nations. Available online: https://www.un.org/en/climatechange/raising-ambition/renewable-energy (accessed on 15 December 2022).
- 2. Alvarez-Diazcomas, A.; Estévez-Bén, A.A.; Rodríguez-Reséndiz, J.; Martínez-Prado, M.A.; Mendiola-Santíbañez, J.D. A novel RC-based architecture for cell equalization in electric vehicles. *Energies* **2020**, *13*, 2349. [CrossRef]
- 3. Mendoza-Varela, I.A.; Alvarez-Diazcomas, A.; Rodriguez-Resendiz, J.; Martinez-Prado, M.A. Modeling and Control of a Phase-Shifted Full-Bridge Converter for a LiFePO4 Battery Charger. *Electronics* **2021**, *10*, 2568. [CrossRef]
- 4. Alvarez-Diazcomas, A.; Estévez-Bén, A.A.; Rodríguez-Reséndiz, J.; Martínez-Prado, M.A.; Carrillo-Serrano, R.V.; Thenozhi, S. A review of battery equalizer circuits for electric vehicle applications. *Energies* **2020**, *13*, 5688. [CrossRef]
- 5. Alvarez-Diazcomas, A.; Rodríguez-Reséndiz, J.; Carrillo-Serrano, R.V. An Improved Battery Equalizer with Reduced Number of Components Applied to Electric Vehicles. *Batteries* 2023, *9*, 65. [CrossRef]
- 6. Benefits of Electric Cars on Environment | EV & Petrol Cars | EDF. Available online: edfenergy.com (accessed on 15 December 2022).
- 7. Johansson, B. Security aspects of future renewable energy systems—A short overview. *Energy* **2013**, *61*, 598–605. [CrossRef]
- Phuangpornpitak, N.; Tia, S. Opportunities and challenges of integrating renewable energy in smart grid system. *Energy Procedia* 2013, 34, 282–290. [CrossRef]
- 9. Basit, M.A.; Dilshad, S.; Badar, R.; Sami ur Rehman, S.M. Limitations, challenges, and solution approaches in grid-connected renewable energy systems. *Int. J. Energy Res.* **2020**, *44*, 4132–4162. [CrossRef]
- Deb, S.; Kalita, K.; Mahanta, P. Review of impact of electric vehicle charging station on the power grid. In Proceedings of the 2017 International Conference on Technological Advancements in Power and Energy (TAP Energy), Kollam, India, New York, NY, USA, 21–23 December 2017; pp. 1–6.
- 11. Deb, S.; Kalita, K.; Mahanta, P. Distribution network planning considering the impact of electric vehicle charging station load. In *Smart Power Distribution Systems*; Academic Press: Cambridge, MA, USA, 2019; pp. 529–553.
- 12. Sachan, S.; Deb, S.; Singh, S.N.; Singh, P.P.; Sharma, D.D. Planning and operation of EV charging stations by chicken swarm optimization driven heuristics. *Energy Convers. Econ.* **2021**, *2*, 91–99. [CrossRef]
- 13. Kumar, R.R.; Alok, K. Adoption of electric vehicle: A literature review and prospects for sustainability. *J. Clean. Prod.* 2020, 253, 119911. [CrossRef]
- 14. Chen, T.; Zhang, X.P.; Wang, J.; Li, J.; Wu, C.; Hu, M.; Bian, H. A review on electric vehicle charging infrastructure development in the UK. J. Mod. Power Syst. Clean Energy 2020, 8, 193–205. [CrossRef]
- 15. Mohammad, A.; Zamora, R.; Lie, T.T. Integration of electric vehicles in the distribution network: A review of PV based electric vehicle modelling. *Energies* 2020, *13*, 4541. [CrossRef]
- Calearo, L.; Marinelli, M.; Ziras, C. A review of data sources for electric vehicle integration studies. *Renew. Sustain. Energy Rev.* 2021, 151, 111518. [CrossRef]
- 17. Deb, S.; Pihlatie, M.; Al-Saadi, M. Smart Charging: A Comprehensive Review. IEEE Access. 2022, 10, 134690–134703. [CrossRef]
- 18. Ismail, A.A.; Mbungu, N.T.; Elnady, A.; Bansal, R.C.; Hamid, A.K.; AlShabi, M. Impact of electric vehicles on smart grid and future predictions: A survey. *Int. J. Model. Simul.* **2022**, 1–17. [CrossRef]
- 19. Sachan, S.; Deb, S.; Singh, P.P.; Alam, M.S.; Shariff, S.M. A comprehensive review of standards and best practices for utility grid integration with electric vehicle charging stations. *Wiley Interdiscip. Rev. Energy Environ.* **2022**, *11*, e424. [CrossRef]
- Deb, S.; Sachan, S.; Alam, M.S.; Shariff, S.M. Electric Vehicle Integrated Virtual Power Plants: A Systematic Review. In Smart Charging Solutions for Hybrid and Electric Vehicles; Wiley Online Library: Hoboken, NJ, USA, 2022; pp. 361–379.
- 21. Deb, S. Machine Learning for Solving Charging Infrastructure Planning Problems: A Comprehensive Review. *Energies* **2021**, *14*, 7833. [CrossRef]
- Deb, S.; Alam, S. Comprehensive Review of Planning Models for Charging Station Placement. In Proceedings of the 2021 5th International Conference on Smart Grid and Smart Cities (ICSGSC), Tokyo, Japan, 18–20 June 2021; IEEE: New York, NY, USA, 2021; pp. 33–38.
- 23. Deb, S. Planning of Sustainable Charging Infrastructure for Smart Cities. In *Flexible Resources for Smart Cities*; Springer: Cham, Switzerland, 2021; pp. 115–132.
- 24. Worighi, I.; Maach, A.; Hafid, A.; Hegazy, O.; Van Mierlo, J. Integrating renewable energy in smart grid system: Architecture, virtualization and analysis. *Sustain. Energy Grids Netw.* **2019**, *18*, 100226. [CrossRef]
- 25. Impram, S.; Nese, S.V.; Oral, B. Challenges of renewable energy penetration on power system flexibility: A survey. *Energy Strategy Rev.* **2020**, *31*, 100539. [CrossRef]
- 26. Nwaigwe, K.N.; Mutabilwa, P.; Dintwa, E. An overview of solar power (PV systems) integration into electricity grids. *Mater. Sci. Energy Technol.* **2019**, *2*, 629–633. [CrossRef]
- 27. Hache, E.; Palle, A. Renewable energy source integration into power networks, research trends and policy implications: A bibliometric and research actors survey analysis. *Energy Policy* **2019**, *124*, 23–35. [CrossRef]
- 28. Heptonstall, P.J.; Gross, R.J. A systematic review of the costs and impacts of integrating variable renewables into power grids. *Nat. Energy* **2021**, *6*, 72–83. [CrossRef]
- 29. Bajaj, M.; Singh, A.K. Grid integrated renewable DG systems: A review of power quality challenges and state-of-the-art mitigation techniques. *Int. J. Energy Res.* 2020, 44, 26–69. [CrossRef]

- 30. Mararakanye, N.; Bekker, B. Renewable energy integration impacts within the context of generator type, penetration level and grid characteristics. *Renew. Sustain. Energy Rev.* **2019**, *108*, 441–451. [CrossRef]
- Deng, X.; Lv, T. Power system planning with increasing variable renewable energy: A review of optimization models. J. Clean. Prod. 2020, 246, 118962. [CrossRef]
- Ahmed, S.D.; Al-Ismail, F.S.; Shafiullah, M.; Al-Sulaiman, F.A.; El-Amin, I.M. Grid integration challenges of wind energy: A review. *IEEE Access* 2020, *8*, 10857–10878. [CrossRef]
- 33. Kebede, A.A.; Kalogiannis, T.; Van Mierlo, J.; Berecibar, M. A comprehensive review of stationary energy storage devices for large scale renewable energy sources grid integration. *Renew. Sustain. Energy Rev.* **2022**, *159*, 112213. [CrossRef]
- Gawusu, S.; Zhang, X.; Ahmed, A.; Jamatutu, S.A.; Miensah, E.D.; Amadu, A.A.; Osei, F.A.J. Renewable energy sources from the perspective of blockchain integration: From theory to application. *Sustain. Energy Technol. Assess.* 2022, 52, 102108. [CrossRef]
- 35. Ranjan, M.; Shankar, R. A literature survey on load frequency control considering renewable energy integration in power system: Recent trends and future prospects. *J. Energy Storage* **2022**, *45*, 103717. [CrossRef]
- Kabeyi, M.J.B.; Olanrewaju, O.A. Sustainable energy transition for renewable and low carbon grid electricity generation and supply. Front. Energy Res. 2022, 9, 743114. [CrossRef]
- Electric Car Registrations and Sales Share in Selected Countries, 2016–2021—Charts—Data & Statistics—IEA. Available online: https://www.iea.org/data-and-statistics/charts/electric-car-registrations-and-sales-share-in-selected-countries-2016-2021 (accessed on 24 January 2023).
- Electric Car Statistics—EV Data [Update: Jan 23] | Heycar. Available online: https://heycar.co.uk/blog/electric-cars-statisticsand-projections (accessed on 24 January 2023).
- Trends in Charging Infrastructure—Global EV Outlook 2022—Analysis—IEA. Available online: https://www.iea.org/reports/ global-ev-outlook-2022/trends-in-charging-infrastructure (accessed on 24 January 2023).
- Electric LDV Per Charging Point in Selected Countries, 2010–2021—Charts—Data & Statistics—IEA. Available online: https: //www.iea.org/data-and-statistics/charts/electric-ldv-per-charging-point-in-selected-countries-2010-2021-2 (accessed on 24 January 2023).
- Renewable Energy—Our World in Data. Available online: https://ourworldindata.org/renewable-energy (accessed on 24 January 2023).
- 42. Kazemzadeh, E.; Koengkan, M.; Fuinhas, J.A. Effect of Battery-Electric and Plug-In Hybrid Electric Vehicles on PM2. 5 Emissions in 29 European Countries. *Sustainability* **2022**, *14*, 2188. [CrossRef]
- 43. Suttakul, P.; Wongsapai, W.; Fongsamootr, T.; Mona, Y.; Poolsawat, K. Total cost of ownership of internal combustion engine and electric vehicles: A real-world comparison for the case of Thailand. *Energy Rep.* **2022**, *8*, 545–553. [CrossRef]
- 44. Nadeem, A.; Rossi, M.; Corradi, E.; Jin, L.; Comodi, G.; Sheikh, N.A. Energy-Environmental Planning of Electric Vehicles (EVs): A Case Study of the National Energy System of Pakistan. *Energies* 2022, *15*, 3054. [CrossRef]
- 45. Wesseh, P.K.; Benjamin, N.I.; Lin, B. The coordination of pumped hydro storage, electric vehicles, and climate policy in imperfect electricity markets: Insights from China. *Renew. Sustain. Energy Rev.* **2022**, *160*, 112275. [CrossRef]
- Li, Y.; Taghizadeh-Hesary, F. The economic feasibility of green hydrogen and fuel cell electric vehicles for road transport in China. Energy Policy 2022, 160, 112703. [CrossRef]
- Asadi, S.; Nilashi, M.; Samad, S.; Abdullah, R.; Mahmoud, M.; Alkinani, M.H.; Yadegaridehkordi, E. Factors impacting consumers' intention toward adoption of electric vehicles in Malaysia. J. Clean. Prod. 2021, 282, 124474. [CrossRef]
- 48. Schulz, F.; Rode, J. Public charging infrastructure and electric vehicles in Norway. Energy Policy 2022, 160, 112660. [CrossRef]
- 49. Toklu, E. Overview of potential and utilization of renewable energy sources in Turkey. Renew. Energy 2013, 50, 456–463. [CrossRef]
- 50. Poudyal, R.; Loskot, P.; Nepal, R.; Parajuli, R.; Khadka, S.K. Mitigating the current energy crisis in Nepal with renewable energy sources. *Renew. Sustain. Energy Rev.* 2019, 116, 109388. [CrossRef]
- 51. Shakeel, S.R.; Takala, J.; Shakeel, W. Renewable energy sources in power generation in Pakistan. *Renew. Sustain. Energy Rev.* 2016, 64, 421–434. [CrossRef]
- Ślusarczyk, B.; Żegleń, P.; Kluczek, A.; Nizioł, A.; Górka, M. The impact of renewable energy sources on the economic growth of Poland and Sweden considering COVID-19 times. *Energies* 2022, 15, 332. [CrossRef]
- 53. Ahmad, U.S.; Usman, M.; Hussain, S.; Jahanger, A.; Abrar, M. Determinants of renewable energy sources in Pakistan: An overview. *Environ. Sci. Pollut. Res.* 2022, 29, 29183–29201. [CrossRef] [PubMed]
- 54. Pereira Jr, A.O.; da Costa, R.C.; do Vale Costa, C.; de Moraes Marreco, J.; La Rovere, E.L. Perspectives for the expansion of new renewable energy sources in Brazil. *Renew. Sustain. Energy Rev.* **2013**, *23*, 49–59. [CrossRef]
- 55. Barbounis, T.; Theocharis, J.; Alexiadis, M.; Dokopoulos, P. Long-term wind speed and power forecasting using local recurrent neural network models. *IEEE Trans. Energy Convers.* 2006, 21, 273–284. [CrossRef]
- Bhaskar, K.; Singh, S. AWNN-Assisted wind power forecasting using feed-forward neural network. *IEEE Trans. Sustain. Energy* 2012, 3, 306–315. [CrossRef]
- Bracale, A.; Caramia, P.; Carpinelli, G.; DiFazi, A.; Varilone, P. A bayesian-based approach for a short-term steady-state forecast of a smart grid. *IEEE Trans. Smart Grid* 2012, 4, 1760–1771. [CrossRef]
- 58. Chang, W. A literature review of wind forecasting methods. J. Power Energy Eng. 2012, 2, 161–168. [CrossRef]
- Deb, S.; Gao, X.Z. Prediction of Charging Demand of Electric City Buses of Helsinki, Finland by Random Forest. *Energies* 2022, 15, 3679. [CrossRef]

- 60. Xydas, E.; Marmaras, C.; Cipcigan, L.M.; Jenkins, N.; Carroll, S.; Barker, M. A data-driven approach for characterising the charging demand of electric vehicles: A UK case study. *Appl. Energy* **2016**, *162*, 763–771. [CrossRef]
- 61. Li, X.; Wang, Z.; Zhang, L.; Sun, F.; Cui, D.; Hecht, C.; Figgener, J.; Sauer, D.U. Electric vehicle behavior modeling and applications in vehicle-grid integration: An overview. *Energy* **2023**, *268*, 126647. [CrossRef]
- 62. Zhan, W.; Wang, Z.; Zhang, L.; Liu, P.; Cui, D.; Dorrell, D.G. A review of siting, sizing, optimal scheduling, and cost-benefit analysis for battery swapping stations. *Energy* **2022**, 258, 124723. [CrossRef]
- 63. Bae, S.; Kwasinski, A. Spatial and temporal model of electric vehicle charging demand. *IEEE Trans. Smart Grid* 2012, *3*, 394–403. [CrossRef]
- 64. Balram, P.; LeAnh, T.; Bertling Tjernberg, L. Effects of plug-in electric vehicle charge scheduling on the day-ahead electricity market price. In Proceedings of the 3rd IEEE PES International Conference and Exhibition on Innovative Smart Grid Technologies, Berlin, Germany, 14–17 October 2012; pp. 1–8.
- 65. DeCraemer, K.; Vandael, S.; Claessens, B.; Deconinck, G. Anevent-driven dual coordination mechanism for demand side management of PHEVs. *IEEE Trans. Smart Grid* 2014, *5*, 751–760. [CrossRef]
- 66. Xie, L.; Carvalho, P.M.; Ferreira, L.A.; Liu, J.; Krogh, B.H.; Popli, N.; Ilić, M.D. Wind integration in power systems: Operational challenges and possible solutions. *Proc. IEEE* **2011**, *99*, 214–232. [CrossRef]
- 67. Sexauer, J.M.; Mohagheghi, S. Voltage quality assessment in a distribution system with distributed generation—A probabilistic load flow approach. *IEEE Trans. Power Del.* 2013, *28*, 1652–1662. [CrossRef]
- Ghosh, S.; Kalagnanam, J.; Katz, D.; Squillante, M.; Zhang, X. Integration of demand response and renewable resources for power generation management. In Proceedings of the ISGT 2011, Anaheim, CA, USA, 17–19 January 2011; pp. 1–7.
- 69. Tushar, M.H.K.; Assi, C.; Maier, M.; Uddin, M.F. Smart microgrids: Optimal joint scheduling for electric vehicles and home appliances. *IEEE Trans. Smart Grid.* 2014, *5*, 239–250. [CrossRef]
- Chen, Q.; Liu, N.; Wang, C.; Zhang, J. Optimal power utilizing strategy for PV-based EV charging stations considering real-time price. In Proceedings of the 2014 IEEE Conference and Expo Transportation Electrification Asia-Pacific (ITEC Asia-Pacific), Beijing, China, 31 August–3 September 2014; pp. 1–6.
- 71. Liu, N.; Chen, Q.; Liu, J.; Lu, X.; Li, P.; Lei, J.; Zhang, J. A heuristic operation strategy for commercial building microgrids containing EVs and PV system. *IEEE Trans. Ind. Electron.* **2015**, *62*, 2560–2570. [CrossRef]
- 72. Liu, N.; Chen, Q.; Lu, X.; Liu, J.; Zhang, J. A charging strategy for PV-based battery switch stations considering service availability and self consumption of PV energy. *IEEE Trans. Ind. Electron.* **2015**, *62*, 4878–4889. [CrossRef]
- 73. Jian, L.; Xue, H.; Xu, G.; Zhu, X.; Zhao, D.; Shao, Z. Regulated charging of plug-in hybrid electric vehicles for minimizing load variance in household smart microgrid. *IEEE Trans. Ind. Electron.* **2013**, *60*, 3218–3226. [CrossRef]
- 74. Yang, L.; Zhang, J.; Poor, H.V. Risk-aware day-ahead scheduling and real-time dispatch for electric vehicle charging. *IEEE Trans. Smart Grid* **2014**, *5*, 693–702. [CrossRef]
- 75. Tang, W.; Bi, S.; Zhang, Y.J.A. Online coordinated charging decision algorithm for electric vehicles without future information. *IEEE Trans. Smart Grid* **2014**, *5*, 2810–2824. [CrossRef]
- 76. Huang, Q.; Jia, Q.; Guan, X. Robust Scheduling of EV Charging Load With Uncertain Wind Power Integration. *IEEE Trans. Smart Grid* 2018, *9*, 1043–1054. [CrossRef]
- 77. Jones, C.B.; Vining, W.; Lave, M.; Haines, T.; Neuman, C.; Bennett, J.; Scoffield, D.R. Impact of Electric Vehicle customer response to Time-of-Use rates on distribution power grids. *Energy Rep.* **2022**, *8*, 8225–8235. [CrossRef]
- 78. Kazemtarghi, A.; Chandwani, A.; Ishraq, N.; Mallik, A. Active Compensation-based Harmonic Reduction Technique to Mitigate Power Quality Impacts of EV Charging Systems. *IEEE Trans. Transp. Electrif.* **2022**, *9*, 1629–1640. [CrossRef]
- Singh, R.S.; Mier, G.; Bosma, T.; Eijgelaar, M.; Bloemhof, G.; Sauba, G. Assessment of EV charging strategies and their effect on residential grids using co-simulation. In *Proceedings of the 2022 International Conference on Smart Energy Systems and Technologies* (SEST), Eindhoven, The Netherlands, 5–7 September 2022; IEEE: New York, NY, USA, 2022; pp. 1–6.
- 80. Qiu, Y.L.; Wang, Y.D.; Iseki, H.; Shen, X.; Xing, B.; Zhang, H. Empirical grid impact of in-home electric vehicle charging differs from predictions. *Resour. Energy Econ.* 2022, 67, 101275. [CrossRef]
- 81. Pankaj, S.; Khalid, M.R.; Alam, M.S.; Asghar, M.J.; Hameed, S. Electric Vehicle Charging Stations and their Impact on the Power Quality of Utility Grid. In *Proceedings of the 2022 International Conference on Decision Aid Sciences and Applications (DASA), Chiangrai, Thailand, 23–25 October 2022*; IEEE: New York, NY, USA, 2022; pp. 816–821.
- 82. Alquthami, T.; Alsubaie, A.; Alkhraijah, M.; Alqahtani, K.; Alshahrani, S.; Anwar, M. Investigating the Impact of Electric Vehicles Demand on the Distribution Network. *Energies* **2022**, *15*, 1180. [CrossRef]
- 83. Ding, Y.; Li, X.; Jian, S. Modeling the impact of vehicle-to-grid discharge technology on transport and power systems. *Transp. Res. Part D Transp. Environ.* **2022**, 105, 103220. [CrossRef]
- 84. Rancilio, G.; Bovera, F.; Delfanti, M. Tariff-based regulatory sandboxes for EV smart charging: Impacts on the tariff and the power system in a national framework. *Int. J. Energy Res.* 2022, *46*, 14794–14813. [CrossRef]
- 85. Rahman, S.; Khan, I.A.; Khan, A.A.; Mallik, A.; Nadeem, M.F. Comprehensive review & impact analysis of integrating projected electric vehicle charging load to the existing low voltage distribution system. *Renew. Sustain. Energy Rev.* **2022**, *153*, 111756.
- Mane, J.A.; Rade, M.R.; Buwa, O.N. Impacts of Electric Vehicle Charger on the Power Grid. In *ISUW 2019*; Springer: Singapore, 2022; pp. 91–100.

- 87. Deb, S.; Sachan, S. Proposed Power Systems Planning in Indian Scenario for Integrating EV Charging Infrastructure. In *Developing Charging Infrastructure and Technologies for Electric Vehicles*; IGI Global: Hershey, PA, USA, 2022; pp. 25–37.
- Zaferanlouei, S.; Lakshmanan, V.; Bjarghov, S.; Farahmand, H.; Korpås, M. BATTPOWER application: Large-scale integration of EVs in an active distribution grid–A Norwegian case study. *Electr. Power Syst. Res.* 2022, 209, 107967. [CrossRef]
- Sachan, S.; Kumar, L.; Deb, S. Smart Charging of Electric Vehicles Considering User Behavior. In Proceedings of the 2021 13th IEEE PES Asia Pacific Power & Energy Engineering Conference (APPEEC), Kerala, India, 21–23 November 2021; IEEE: New York, NY, USA, 2021; pp. 1–4.
- 90. Dechanupaprittha, S.; Jamroen, C. Self-learning PSO based optimal EVs charging power control strategy for frequency stabilization considering frequency deviation and impact on EV owner. *Sustain. Energy Grids Netw.* **2021**, *26*, 100463. [CrossRef]
- Ahmed, A.; Iqbal, A.; Khan, I.; Al-Wahedi, A.; Mehrjerdi, H.; Rahman, S. Impact of EV charging station penetration on harmonic distortion level in utility distribution network: A case study of Qatar. In *Proceedings of the 2021 IEEE Texas Power and Energy Conference (TPEC), College Station, TX, USA, 2–5 February 2021;* IEEE: New York, NY, USA, 2021; pp. 1–6.
- Khan, M.M.H.; Hossain, A.; Ullah, A.; Hossain Lipu, M.S.; Siddiquee, S.S.; Alam, M.S.; Jamal, T.; Ahmed, H. Integration of large-scale electric vehicles into utility grid: An efficient approach for impact analysis and power quality assessment. *Sustainability* 2021, 13, 10943. [CrossRef]
- Venegas, F.G.; Petit, M.; Perez, Y. Plug-in behavior of electric vehicles users: Insights from a large-scale trial and impacts for grid integration studies. *eTransportation* 2021, 10, 100131. [CrossRef]
- 94. Qiu, K.; Naim, W.; Shayesteh, E.; Hilber, P. Reliability evaluation of power distribution grids considering the dynamic charging mode of electric buses. *Energy Rep.* **2021**, *7*, 134–140. [CrossRef]
- 95. Mowry, A.M.; Mallapragada, D.S. Grid impacts of highway electric vehicle charging and role for mitigation via energy storage. *Energy Policy* **2021**, *157*, 112508. [CrossRef]
- 96. Stiasny, J.; Zufferey, T.; Pareschi, G.; Toffanin, D.; Hug, G.; Boulouchos, K. Sensitivity analysis of electric vehicle impact on low-voltage distribution grids. *Electr. Power Syst. Res.* **2021**, *191*, 106696. [CrossRef]
- Soofi, A.F.; Bayani, R.; Manshadi, S.D. Analyzing power quality implications of high level charging rates of electric vehicle within distribution networks. In Proceedings of the 2021 IEEE Transportation Electrification Conference & Expo (ITEC), Online, New York, NY, USA, 2–25 June 2021; pp. 684–689.
- Iqbal, M.N.; Kütt, L.; Daniel, K.; Asad, B.; Shams Ghahfarokhi, P. Estimation of harmonic emission of electric vehicles and their impact on low voltage residential network. *Sustainability* 2021, 13, 8551. [CrossRef]
- Slangen, T.; Bhattacharyya, S. Determining the impacts of fast-charging of electric buses on the power quality based on field measurements. In Proceedings of the CIRED 2021-The 26th International Conference and Exhibition on Electricity Distribution, Online, 20–23 September 2021; Volume 2021, pp. 778–782.
- Archana, A.N.; Rajeev, T. A Novel Reliability Index Based Approach for EV Charging Station Allocation in Distribution System. *IEEE Trans. Ind. Appl.* 2021, 57, 6385–6394. [CrossRef]
- 101. Akil, M.; Dokur, E.; Bayindir, R. Impact of electric vehicle charging profiles in data-driven framework on distribution network. In Proceedings of the 2021 9th International Conference on Smart Grid (ICSMARTGRID), Setubal, Portugal, 29 June–1 July 2021; IEEE: New York, NY, USA, 2021; pp. 220–225.
- 102. Van den Berg, M.A.; Lampropoulos, I.; AlSkaif, T.A. Impact of electric vehicles charging demand on distribution transformers in an office area and determination of flexibility potential. *Sustain. Energy Grids Netw.* **2021**, *26*, 100452. [CrossRef]
- Haider, S.; Schegner, P. Simulating the Impacts of Uncontrolled Electric Vehicle Charging in Low Voltage Grids. *Energies* 2021, 14, 2330. [CrossRef]
- 104. Sharma, G.; Sood, V.K.; Alam, M.S.; Shariff, S.M. Comparison of common DC and AC bus architectures for EV fast charging stations and impact on power quality. *ETransportation* **2020**, *5*, 100066. [CrossRef]
- Sachan, S.; Deb, S.; Singh, S.N. Different charging infrastructures along with smart charging strategies for electric vehicles. *Sustain. Cities Soc.* 2020, 60, 102238. [CrossRef]
- Deb, S.; Tammi, K.; Gao, X.Z.; Kalita, K.; Mahanta, P. A hybrid multi-objective chicken swarm optimization and teaching learning based algorithm for charging station placement problem. *IEEE Access* 2020, *8*, 92573–92590. [CrossRef]
- 107. González, L.G.; Siavichay, E.; Espinoza, J.L. Impact of EV fast charging stations on the power distribution network of a Latin American intermediate city. *Renew. Sustain. Energy Rev.* **2019**, *107*, 309–318. [CrossRef]
- Lillebo, M.; Zaferanlouei, S.; Zecchino, A.; Farahmand, H. Impact of large-scale EV integration and fast chargers in a Norwegian LV grid. J. Eng. 2019, 2019, 5104–5108. [CrossRef]
- Deb, S.; Tammi, K.; Kalita, K.; Mahanta, P. Charging station placement for electric vehicles: A case study of Guwahati city, India. IEEE Access 2019, 7, 100270–100282. [CrossRef]
- Fischer, D.; Harbrecht, A.; Surmann, A.; McKenna, R. Electric vehicles' impacts on residential electric local profiles–A stochastic modelling approach considering socio-economic, behavioural and spatial factors. *Appl. Energy* 2019, 233, 644–658. [CrossRef]
- 111. Deb, S.; Tammi, K.; Kalita, K.; Mahanta, P. Impact of electric vehicle charging station load on distribution network. *Energies* **2018**, *11*, 178. [CrossRef]
- 112. Kongjeen, Y.; Bhumkittipich, K. Impact of plug-in electric vehicles integrated into power distribution system based on voltagedependent power flow analysis. *Energies* **2018**, *11*, 1571. [CrossRef]
- 113. Sachan, S. Stochastic charging of electric vehicles in smart power distribution grids. Sustain. Cities Soc. 2018, 40, 91–100. [CrossRef]

- Lin, H.; Liu, Y.; Sun, Q.; Xiong, R.; Li, H.; Wennersten, R. The impact of electric vehicle penetration and charging patterns on the management of energy hub–A multi-agent system simulation. *Appl. Energy* 2018, 230, 189–206. [CrossRef]
- Deb, S.; Kalita, K.; Mahanta, P. Impact of electric vehicle charging stations on reliability of distribution network. In Proceedings of the 2017 International Conference on Technological Advancements in Power and Energy (TAP Energy), Kollam, India, New York, NY, USA, 21–23 December 2017; pp. 1–6.
- 116. Wang, Y.; Xu, C.; Yuan, P. Is there a grid-connected effect of grid infrastructure on renewable energy generation? Evidence from China's upgrading transmission lines. *Energy Environ.* **2022**, *33*, 975–995. [CrossRef]
- 117. Yu, S.; Zhou, S.; Qin, J. Layout optimization of China's power transmission lines for renewable power integration considering flexible resources and grid stability. *Int. J. Electr. Power Energy Syst.* **2022**, 135, 107507. [CrossRef]
- 118. Bhattacharya, S.; Pennock, S.; Robertson, B.; Hanif, S.; Alam, M.J.E.; Bhatnagar, D.; Preziuso, D.; O'Neil, R. Timing value of marine renewable energy resources for potential grid applications. *Appl. Energy* **2021**, *299*, 117281. [CrossRef]
- 119. Akhtar, I.; Kirmani, S.; Jameel, M. Reliability Assessment of Power System Considering the Impact of Renewable Energy Sources Integration into Grid with Advanced Intelligent Strategies. *IEEE Access* **2021**, *9*, 32485–32497. [CrossRef]
- 120. Wirawan, H.; Gultom, Y. M The effects of renewable energy-based village grid electrification on poverty reduction in remote areas: The case of Indonesia. *Energy Sustain. Dev.* **2021**, *62*, 186–194. [CrossRef]
- 121. Jacobson, M.Z. The cost of grid stability with 100% clean, renewable energy for all purposes when countries are isolated versus interconnected. *Renew. Energy* 2021, *179*, 1065–1075. [CrossRef]
- 122. Mokeke, S.; Thamae, L.Z. The impact of intermittent renewable energy generators on Lesotho national electricity grid. *Electr. Power Syst. Res.* **2021**, *196*, 107196. [CrossRef]
- 123. Ayadi, F.; Colak, I.; Garip, I.; Bulbul, H.I. Impacts of Renewable Energy Resources in Smart Grid. In Proceedings of the 2020 8th International Conference on Smart Grid (icSmartGrid), Paris, France, 17–19 June 2020; pp. 183–188. [CrossRef]
- 124. Oyekale, J.; Petrollese, M.; Tola, V.; Cau, G. Impacts of Renewable Energy Resources on Effectiveness of Grid-Integrated Systems: Succinct Review of Current Challenges and Potential Solution Strategies. *Energies* **2020**, *13*, 4856. [CrossRef]
- 125. Alam, M.S.; Al-Ismail, F.; Salem, A.; Abido, M.A. High-Level Penetration of Renewable Energy Sources into Grid Utility: Challenges and Solutions. *IEEE Access* 2020, *8*, 190277–190299. [CrossRef]
- 126. Østergaard, P.A.; Duic, N.; Noorollahi, Y.; Mikulcic, H.; Kalogirou, S. Sustainable development using renewable energy technology. *Renew. Energy* 2020, 146, 2430–2437. [CrossRef]
- 127. Seck, G.S.; Krakowski, V.; Assoumou, E.; Maïzi, N.; Mazauric, V. Embedding power system's reliability within a long-term Energy System Optimization Model: Linking high renewable energy integration and future grid stability for France by 2050. *Appl. Energy* 2020, 257, 114037. [CrossRef]
- 128. Al-Shetwi, A.Q.; Hannan, M.A.; Jern, K.P.; Mansur, M.; Mahlia, T.M.I. Grid-connected renewable energy sources: Review of the recent integration requirements and control methods. *J. Clean. Prod.* 2020, 253, 119831. [CrossRef]
- 129. Johnson, S.C.; Rhodes, J.D.; Webber, M.E. Understanding the impact of non-synchronous wind and solar generation on grid stability and identifying mitigation pathways. *Appl. Energy* **2020**, *262*, 114492. [CrossRef]
- Keck, F.; Lenzen, M.; Vassallo, A.; Li, M. The impact of battery energy storage for renewable energy power grids in Australia. *Energy* 2019, 173, 647–657. [CrossRef]
- 131. Johnson, S.C.; Papageorgiou, D.J.; Mallapragada, D.S.; Deetjen, T.A.; Rhodes, J.D.; Webber, M.E. Evaluating rotational inertia as a component of grid reliability with high penetrations of variable renewable energy. *Energy* **2019**, *180*, 258–271. [CrossRef]
- 132. McPherson, M.; Tahseen, S. Deploying storage assets to facilitate variable renewable energy integration: The impacts of grid flexibility, renewable penetration, and market structure. *Energy* **2018**, 145, 856–870. [CrossRef]
- 133. Perera, A.T.D.; Nik, V.M.; Mauree, D.; Scartezzini, J.L. Electrical hubs: An effective way to integrate non-dispatchable renewable energy sources with minimum impact to the grid. *Appl. Energy* **2017**, *190*, 232–248. [CrossRef]
- 134. Musau, M.P.; Chepkania, T.; Odero, A.; Wekesa, C.W. Effects of renewable energy on frequency stability: A proposed case study of the Kenyan grid. In Proceedings of the 2017 IEEE PES Power Africa, Accra, Ghana, 27–30 June 2017; pp. 12–15. [CrossRef]
- 135. Triviño-Cabrera, A.; Longo, M.; Foiadelli, F. Impact of renewable energy sources in the power quality of the Italian electric grid. In Proceedings of the 2017 11th IEEE International Conference on Compatibility, Power Electronics and Power Engineering (CPE-POWERENG), Cadiz, Spain, 4–6 April 2017; pp. 576–581. [CrossRef]
- 136. Harrouz, A.; Abbes, M.; Colak, I.; Kayisli, K. Smart grid and renewable energy in Algeria. In Proceedings of the 2017 IEEE 6th International Conference on Renewable Energy Research and Applications (ICRERA), San Diego, CA, USA, 5–8 November 2017; pp. 1166–1171. [CrossRef]
- 137. Schmietendorf, K.; Peinke, J.; Kamps, O. The impact of turbulent renewable energy production on power grid stability and quality. *Eur. Phys. J. B* **2017**, *90*, 222. [CrossRef]
- Electric Vehicle Integration into Power Grids (entsoe.eu). Available online: https://www.entsoe.eu/2021/04/02/electric-vehicleintegration-into-power-grids/ (accessed on 10 January 2023).
- 139. Telukunta, V.; Pradhan, J.; Agrawal, A.; Singh, M.; Srivani, S.G. Protection challenges under bulk penetration of renewable energy resources in power systems: A review. *CSEE J. Power Energy Syst.* **2017**, *3*, 365–379. [CrossRef]
- Zhang, L.; Yin, Q.; Zhang, Z.; Zhu, Z.; Lyu, L.; Hai, K.L.; Cai, G. A wind power curtailment reduction strategy using electric vehicles based on individual differential evolution quantum particle swarm optimization algorithm. *Energy Rep.* 2022, *8*, 14578–14594. [CrossRef]

- 141. Huy, T.H.B.; Dinh, H.T.; Kim, D. Multi-objective framework for a home energy management system with the integration of solar energy and an electric vehicle using an augmented ε-constraint method and lexicographic optimization. *Sustain. Cities Soc.* 2023, 88, 104289. [CrossRef]
- 142. Li, Y.; Wang, F.; Tang, X.; Hu, X.; Lin, X. Convex optimization-based predictive and bi-level energy management for plug-in hybrid electric vehicles. *Energy* 2022, 257, 124672. [CrossRef]
- 143. Wang, W.; Chen, Y.; Yang, C.; Li, Y.; Xu, B.; Xiang, C. An enhanced hypotrochoid spiral optimization algorithm based intertwined optimal sizing and control strategy of a hybrid electric air-ground vehicle. *Energy* **2022**, 257, 124749. [CrossRef]
- 144. Yang, W.; Guo, J.; Vartosh, A. Optimal economic-emission planning of multi-energy systems integrated electric vehicles with modified group search optimization. *Appl. Energy* **2022**, *311*, 118634. [CrossRef]
- 145. Vasanthkumar, P.; Revathi, A.R.; Devi, G.R.; Kavitha, R.J.; Muniappan, A.; Karthikeyan, C. Improved wild horse optimizer with deep learning enabled battery management system for internet of things based hybrid electric vehicles. *Sustain. Energy Technol. Assess.* **2022**, *52*, 102281. [CrossRef]
- 146. Nayak, M.R.; Behura, D.; Nayak, S. Performance analysis of unbalanced radial feeder for integrating energy storage system with wind generator using inherited competitive swarm optimization algorithm. *J. Energy Storage* **2021**, *38*, 102574. [CrossRef]
- Wang, H.Y.; Chen, B.; Pan, D.; Lv, Z.A.; Huang, S.Q.; Khayatnezhad, M.; Jimenez, G. Optimal wind energy generation considering climatic variables by Deep Belief network (DBN) model based on modified coot optimization algorithm (MCOA). *Sustain. Energy Technol. Assess.* 2022, 53, 102744. [CrossRef]
- Xian, H.; Che, J. Unified whale optimization algorithm based multi-kernel SVR ensemble learning for wind speed forecasting. *Appl. Soft Comput.* 2022, 130, 109690. [CrossRef]
- Ullah, K.; Hafeez, G.; Khan, I.; Jan, S.; Javaid, N. A multi-objective energy optimization in smart grid with high penetration of renewable energy sources. *Appl. Energy* 2021, 299, 117104. [CrossRef]
- 150. Liu, Z.F.; Li, L.L.; Liu, Y.W.; Liu, J.Q.; Li, H.Y.; Shen, Q. Dynamic economic emission dispatch considering renewable energy generation: A novel multi-objective optimization approach. *Energy* **2021**, 235, 121407. [CrossRef]
- Rawa, M.; AlKubaisy, Z.M.; Alghamdi, S.; Refaat, M.M.; Ali, Z.M.; Aleem, S.H.A. A techno-economic planning model for integrated generation and transmission expansion in modern power systems with renewables and energy storage using hybrid Runge Kutta-gradient-based optimization algorithm. *Energy Rep.* 2022, *8*, 6457–6479. [CrossRef]
- Krishnan, V.A.; Kumar, N.S. Robust soft computing control algorithm for sustainable enhancement of renewable energy sources based microgrid: A hybrid Garra rufa fish optimization–Isolation forest approach. *Sustain. Comput. Inform. Syst.* 2022, 35, 100764. [CrossRef]
- 153. Jiang, W.; Wang, X.; Huang, H.; Zhang, D.; Ghadimi, N. Optimal economic scheduling of microgrids considering renewable energy sources based on energy hub model using demand response and improved water wave optimization algorithm. *J. Energy Storage* **2022**, *55*, 105311. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.